UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-O

[x] QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended March 31, 2018

OR

[] TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 For the transition period from to Commission File Number 1-5823

CNA FINANCIAL CORPORATION

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation or organization)

36-6169860 (I.R.S. Employer Identification No.)

333 S. Wabash Chicago, Illinois (Address of principal executive offices)

60604 (Zip Code)

(312) 822-5000

(Registrant's telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days.

Yes [x] No []

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (\$232,405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files).

Yes [x] No []

Indicate by check mark whether the registrant is a large accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company" and "emerging growth company" in Rule 12b-2 of the Exchange Act.

Large accelerated	Accelerated filer []	Non-accelerated	Smaller reporting company [Emerging growth company [
filer [x]		filer [] (Do not check if a]]
		smaller reporting company)		

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. []

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes [] No [x]

Indicate the number of shares outstanding of each of the issuer's classes of common stock, as of the latest practicable date.

Class	Outstanding at April 26, 2018
Common Stock, Par value \$2.50	271,372,358

Item Number		Page Number
	PART I . Financial Information	
1.	Condensed Consolidated Financial Statements:	
	Condensed Consolidated Statements of Operations for the three months ended March 31, 2018 and 2017 (Unaudited)	<u>3</u>
	Condensed Consolidated Statements of Comprehensive Income (Loss) for the three months ended March 31, 2018 and 2017 (Unaudited)	<u>4</u>
	Condensed Consolidated Balance Sheets as of March 31, 2018 (Unaudited) and December 31, 2017	<u>5</u>
	Condensed Consolidated Statements of Cash Flows for the three months ended March 31, 2018 and 2017 (Unaudited)	<u>6</u>
	Condensed Consolidated Statements of Stockholders' Equity for the three months ended March 31, 2018 and 2017 (Unaudited)	<u>8</u>
	Notes to Condensed Consolidated Financial Statements (Unaudited)	<u>9</u>
2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>39</u>
3.	Quantitative and Qualitative Disclosures About Market Risk	<u>56</u>
4.	Controls and Procedures	<u>56</u>
	PART I I. Other Information	
1.	<u>Legal Proceedings</u>	<u>57</u>
6.	<u>Exhibits</u>	<u>57</u>
	2	

PART I. Financial Information

Item 1. Condensed Consolidated Financial Statements

CNA Financial Corporation

Condensed Consolidated Statements of Operations (Unaudited)

Three months ended March 31

(In millions, except per share data)		2018		2017
Revenues				
Net earned premiums	\$	1,785	\$	1,645
Net investment income		490		545
Net realized investment gains				
Other-than-temporary impairment losses		(6)		(2)
Other net realized investment gains		18		38
Net realized investment gains		12		36
Non-insurance warranty revenue (Note J)		238		93
Other revenues		10		11
Total revenues		2,535		2,330
Claims, Benefits and Expenses				
Insurance claims and policyholders' benefits		1,339		1,293
Amortization of deferred acquisition costs		296		305
Non-insurance warranty expense (Note J)		216		70
Other operating expenses		303		276
Interest		35		43
Total claims, benefits and expenses		2,189		1,987
Income before income tax		346		343
Income tax expense		(55)		(83)
Net income	\$	291	\$	260
Basic earnings per share	\$	1.07	\$	0.96
Diluted earnings per share	\$	1.07	\$	0.96
Dividends declared per share	\$	2.30	\$	2.25
Dividends deciated per share	<u> </u>	2.30	Ψ	2.23
Weighted Average Outstanding Common Stock and Common Stock Equivalents				
Basic		271.4		270.7
Diluted		272.4		271.7

CNA Financial Corporation Condensed Consolidated Statements of Comprehensive Income (Loss) (Unaudited)

Three months ended March 31

(In millions)	2018	2017
Comprehensive (Loss) Income		
Net income	\$ 291	\$ 260
Other Comprehensive (Loss) Income, Net of Tax		
Changes in:		
Net unrealized gains on investments with other-than-temporary impairments	(9)	(4)
Net unrealized gains on other investments	(429)	67
Net unrealized gains on investments	(438)	63
Foreign currency translation adjustment	12	11
Pension and postretirement benefits	10	7
Other comprehensive (loss) income, net of tax		81
Total comprehensive (loss) income	\$ (125)	\$ 341

CNA Financial Corporation Condensed Consolidated Balance Sheets

(In millions, except share data)	rch 31, 2018 Jnaudited)	Dec	ember 31, 2017
Assets			
Investments:			
Fixed maturity securities at fair value (amortized cost of \$37,991and \$38,215)	\$ 40,259	\$	41,487
Equity securities at fair value (cost of \$749 and \$659)	770		695
Limited partnership investments	2,364		2,369
Other invested assets	48		44
Mortgage loans	864		839
Short term investments	1,230		1,436
Total investments	45,535		46,870
Cash	282		355
Reinsurance receivables (less allowance for uncollectible receivables of \$29 and \$29)	4,408		4,261
Insurance receivables (less allowance for uncollectible receivables of \$43 and \$44)	2,371		2,292
Accrued investment income	414		411
Deferred acquisition costs	665		634
Deferred income taxes	238		137
Property and equipment at cost (less accumulated depreciation of \$253 and \$274)	346		326
Goodwill	149		148
Other assets	3,241		1,133
Total assets	\$ 57,649	\$	56,567
Liabilities			
Insurance reserves:			
Claim and claim adjustment expenses	\$ 22,067	\$	22,004
Unearned premiums	4,256		4,029
Future policy benefits	10,783		11,179
Short term debt	30		150
Long term debt	2,679		2,708
Other liabilities (includes \$147 and \$143 due to Loews Corporation)	6,409		4,253
Total liabilities	46,224		44,323
Commitments and contingencies (Notes C and F)			
Stockholders' Equity			
Common stock (\$2.50 par value; 500,000,000 shares authorized; 273,040,243 shares issued; 271,371,607 and 271,205,390 shares outstanding)	683		683
Additional paid-in capital	2,173		2,175
Retained earnings	9,028		9,414
Accumulated other comprehensive (loss) income	(400)		32
Treasury stock (1,668,636 and 1,834,853 shares), at cost	(59)		(60)
Total stockholders' equity	11,425		12,244
Total liabilities and stockholders' equity	\$ 57,649	\$	56,567

CNA Financial Corporation Condensed Consolidated Statements of Cash Flows (Unaudited)

Three months ended March 31

(In millions)	2018	2017
Cash Flows from Operating Activities		
Net income	\$ 291 \$	260
Adjustments to reconcile net income to net cash flows provided by operating activities:		
Deferred income tax expense	29	72
Trading portfolio activity	(1)	(6)
Net realized investment gains	(12)	(36)
Equity method investees	(2)	38
Net amortization of investments	(15)	(12)
Depreciation and amortization	20	21
Changes in:		
Receivables, net	(215)	89
Accrued investment income	(3)	(26)
Deferred acquisition costs	(29)	(24)
Insurance reserves	311	135
Other assets	(72)	(37)
Other liabilities	(99)	(206)
Other, net	15	14
Total adjustments	(73)	22
Net cash flows provided by operating activities	218	282
Cash Flows from Investing Activities		
Dispositions:		
Fixed maturity securities - sales	2,576	1,359
Fixed maturity securities - maturities, calls and redemptions	531	823
Equity securities	7	16
Limited partnerships	69	57
Mortgage loans	11	3
Purchases:		
Fixed maturity securities	(2,690)	(2,097)
Equity securities	(98)	(7)
Limited partnerships	(62)	(18)
Mortgage loans	(36)	(23)
Change in other investments	(4)	(1)
Change in short term investments	208	271
Purchases of property and equipment	(38)	(30)
Other, net	15	1
Net cash flows provided by investing activities	489	354

Three months ended March 31

(In millions)	 2018		2017
Cash Flows from Financing Activities			
Dividends paid to common stockholders	\$ (624)	\$	(609)
Repayment of debt	(150)		_
Other, net	(7)		
Net cash flows used by financing activities	 (781)		(609)
Effect of foreign exchange rate changes on cash	 1		1
Net change in cash	(73)		28
Cash, beginning of year	 355		271
Cash, end of period	\$ 282	\$	299

CNA Financial Corporation

Condensed Consolidated Statements of Stockholders' Equity (Unaudited)

Three months ended March 31

(In millions)	2018	2017
Common Stock		
Balance, beginning of period	\$ 683	\$ 683
Balance, end of period	683	683
Additional Paid-in Capital		
Balance, beginning of period	2,175	2,173
Stock-based compensation	(2)	(12)
Balance, end of period	2,173	2,161
Retained Earnings		
Balance, beginning of period, as previously reported	9,414	9,359
Cumulative effect adjustments from changes in accounting guidance, net of tax	(50)	
Balance, beginning of period, as adjusted	9,364	9,359
Dividends paid to common stockholders	(627)	(613)
Net income	291	260
Balance, end of period	9,028	9,006
Accumulated Other Comprehensive (Loss)		
Balance, beginning of period, as previously reported	32	(173)
Cumulative effect adjustments from changes in accounting guidance, net of tax	(16)	
Balance, beginning of period, as adjusted	16	(173)
Other comprehensive (loss) income	(416)	81
Balance, end of period	(400)	(92)
Treasury Stock		
Balance, beginning of period	(60)	(73)
Stock-based compensation	1	8
Balance, end of period	(59)	(65)
Total stockholders' equity	\$ 11,425	\$ 11,693

CNA Financial Corporation Notes to Condensed Consolidated Financial Statements (Unaudited)

Note A . General

Basis of Presentation

The Condensed Consolidated Financial Statements include the accounts of CNA Financial Corporation (CNAF) and its subsidiaries. Collectively, CNAF and its subsidiaries are referred to as CNA or the Company. Loews Corporation (Loews) owned approximately 89% of the outstanding common stock of CNAF as of March 31, 2018.

The accompanying Condensed Consolidated Financial Statements have been prepared in conformity with accounting principles generally accepted in the United States of America (GAAP). Intercompany amounts have been eliminated. Certain financial information that is normally included in annual financial statements prepared in accordance with GAAP, including certain financial statement notes, is not required for interim reporting purposes and has been condensed or omitted. These statements should be read in conjunction with the Consolidated Financial Statements and notes thereto included in CNAF's Annual Report on Form 10-K filed with the Securities and Exchange Commission for the year ended December 31, 2017, including the summary of significant accounting policies in Note A. The preparation of Condensed Consolidated Financial Statements in conformity with GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and the disclosure of contingent assets and liabilities at the date of the Condensed Consolidated Financial Statements and the reported amounts of revenues and expenses during the reporting period. Actual results may differ from those estimates.

The interim financial data as of March 31, 2018 and for the three months ended March 31, 2018 and 2017 is unaudited. However, in the opinion of management, the interim data includes all adjustments, including normal recurring adjustments, necessary for a fair statement of the Company's results for the interim periods. The results of operations for the interim periods are not necessarily indicative of the results to be expected for the full year.

Recently Adopted Accounting Standards Updates (ASU)

ASU 2014-09: In May 2014, the Financial Accounting Standards Board (FASB) issued ASU No. 2014-09, *Revenue Recognition (Topic 606): Revenue from Contracts with Customers*. The standard excludes from its scope the accounting for insurance contracts, financial instruments, and certain other agreements that are governed under other GAAP guidance. The updated guidance requires an entity to recognize revenue as performance obligations are met, in an amount that reflects the consideration the entity is entitled to receive for the transfer of the promised goods or services.

On January 1, 2018, the Company adopted the updated guidance using the modified retrospective method applied to all contracts which were not completed as of the date of adoption, with the cumulative effect recognized as an adjustment to the opening balance of Retained earnings. Prior period amounts have not been adjusted and continue to be reported in accordance with the previous accounting guidance.

Under the new guidance, revenue on warranty products and services will be recognized more slowly compared to the historic revenue recognition pattern. In addition, for warranty products in which the Company acts as the principal in the transaction, Non-insurance warranty revenue and Non-insurance warranty expense are increased to reflect the gross amount paid by consumers, including the retail seller's markup which is considered a commission to the Company's agent. This gross-up of revenue and expense also resulted in an increase to Other assets and Other liabilities on the Company's Condensed Consolidated Balance Sheets as the revenue and expense are recognized over the actuarially determined expected claims emergence pattern.

The cumulative effect changes to the Condensed Consolidated Balance Sheet for the adoption of the updated guidance on January 1, 2018 were as follows:

			ments Due to		
(In millions)	Balance as of Adoption of Topic December 31, 2017 606			Balance as of January 1, 2018	
Other assets	\$ 1,133	\$	1,882	\$	3,015
Other liabilities	4,253		1,969		6,222
Deferred income taxes	137		21		158
Retained earnings	9,414		(66)		9,348

The impact of adoption on the Condensed Consolidated Statement of Operations and Balance Sheet as of and for the three months ended March 31, 2018 was as follows:

Three months ended March 31 (In millions)	2018 Prior to Adoption	Effect of Adoption	2018 as Reported
Statement of operations:			
Non-insurance warranty revenue	\$ 101	\$ 137	\$ 238
Total revenues	2,398	137	2,535
Non-insurance warranty expense	78	138	216
Total claims, benefits and expenses	2,051	138	2,189
Income before income tax	347	(1)	346
Income tax expense	(55)	_	(55)
Net income	292	(1)	291
Balance sheet (1):			
Other assets	\$ 3,154	\$ 87	\$ 3,241
Other liabilities	6,321	88	6,409
Deferred income taxes	238	_	238
Retained earnings	9,029	(1)	9,028

^{(1) 2018} Prior to Adoption includes the cumulative effect adjustment at adoption.

See Note J to the Condensed Consolidated Financial Statements for additional information regarding non-insurance revenues from contracts with customers.

ASU 2016-01: In January 2016, the FASB issued ASU No. 2016-01, Financial Instruments - Overall (Subtopic 825-10): Recognition and Measurement of Financial Assets and Financial Liabilities. The updated accounting guidance requires changes to the reporting model for financial instruments. The guidance primarily changes the model for equity securities by requiring changes in the fair value of equity securities (except those accounted for under the equity method of accounting, those without readily determinable fair values and those that result in consolidation of the investee) to be recognized through the income statement. The Company adopted the updated guidance on January 1, 2018 and recognized a cumulative effect adjustment that increased beginning Retained earnings by \$28 million, net of tax. Prior period amounts have not been adjusted and continue to be reported in accordance with the previous accounting guidance.

For the three months ended March 31, 2018, the Company recognized a \$15 million pretax loss within Net realized investment gains (losses) for the change in fair value of non-redeemable preferred stock and less than a \$1 million pretax gain within Net investment income for the change in fair value of common stock as a result of this change. For the three months ended March 31, 2017 a \$1 million decrease in the fair value of common stock and a \$5 million increase in the fair value of non-redeemable preferred stock was recognized in Other comprehensive income. The Company's non-redeemable preferred stock contain characteristics of debt securities, are priced similarly to bonds, and are held primarily for income generation through periodic dividends. While recognition of gains and losses on these securities are no longer discretionary, management does not consider the changes in fair value of non-redeemable preferred stock to be reflective of our primary operations. As such, the changes in the fair value of these securities are recorded through Net realized investments gains (losses). During the first quarter of 2018, the Company increased its common stock with the intention of holding the securities primarily for market appreciation. As such, the changes in the fair value of these securities are recorded through Net investment income.

ASU 2017-07: In March 2017, the FASB issued ASU No. 2017-07, Compensation-Retirement Benefits (Topic 715): Improving the Presentation of Net Periodic Pension Cost and Net Periodic Postretirement Benefit Cost. The updated accounting guidance requires changes to the presentation of the components of net periodic benefit cost on the income statement by requiring service cost to be presented with other employee compensation costs and other components of net periodic pension cost to be presented outside of any subtotal of operating income. The ASU also stipulates that only the service cost component of net benefit cost is eligible for capitalization. The Company adopted the updated guidance effective January 1, 2018. The guidance was applied on a prospective basis for capitalization of service costs and on a retrospective basis for the presentation of the service cost and other components of net periodic benefit costs in the Company's Condensed Consolidated Statements of Operations and in its disclosures. The Company expanded the related footnote disclosure, Note G to the Condensed Consolidated Financial Statements, to disclose the amount of service cost and non-service cost components of net periodic benefit cost and the line items in the Condensed Consolidated Statements of Operations in which such amounts are reported. The change limiting the costs eligible for capitalization is not material to the Company's results of operations or financial position.

ASU 2018-02: In February 2018, the FASB issued ASU No. 2018-02, *Income Statement - Reporting Comprehensive Income (Topic 220): Reclassification of Certain Tax Effects from Accumulated Other Comprehensive Income.* GAAP requires the remeasurement of deferred tax assets and liabilities due to a change in the tax rate to be included in Net income, even if the related income tax effects were originally recognized in Accumulated other comprehensive income (AOCI). The ASU allows a reclassification from AOCI to Retained earnings for stranded tax effects resulting from the new U.S. Federal corporate income tax rate enacted on December 22, 2017. The Company early adopted the updated guidance effective January 1, 2018 and elected to reclassify the stranded income tax effects relating to the reduction in the Federal corporate income tax rate from AOCI to Retained earnings at the beginning of the period of adoption. The net impact of the accounting change resulted in a \$12 million increase in AOCI and a corresponding decrease in Retained earnings. The \$12 million increase in AOCI is comprised of a \$142 million increase in net unrealized gains (losses) on investments partially offset by a \$130 million decrease in unrecognized pension and postretirement benefits.

The Company releases tax effects from AOCI utilizing the security-by-security approach for Net unrealized gains (losses) on investments with Other-than-temporary impairment (OTTI) losses and Net unrealized gains (losses) on other investments. For Pension and postretirement benefits, tax effects from AOCI are released at enacted tax rates based on the pre-tax adjustments to pension liabilities or assets recognized within OCI.

Accounting Standards Pending Adoption

In February 2016, the FASB issued ASU No. 2016-02, *Leases (Topic 842): Accounting for Leases*. The updated accounting guidance requires lessees to recognize on the balance sheet the assets and liabilities for the rights and obligations created by all leases, including those historically accounted for as operating leases. The guidance is effective for interim and annual periods beginning after December 15, 2018. The Company is currently evaluating the effect the updated guidance will have on the Company's financial statements. It is expected that assets and liabilities will increase based on the present value of remaining lease payments for leases in place at the adoption date; however, this is not expected to be material to the Company's results of operations or financial position.

In June 2016, the FASB issued ASU No. 2016-13, *Financial Instruments-Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments.* The updated accounting guidance requires changes to the recognition of credit losses on financial instruments not accounted for at fair value through net income. The guidance is effective for interim and annual periods beginning after December 15, 2019. The Company is currently evaluating the effect the guidance will have on the Company's financial statements, but expects the primary changes to be the use of the expected credit loss model for its mortgage loan portfolio and reinsurance receivables and the presentation of credit losses within the available-for-sale fixed maturities portfolio through an allowance method rather than as a direct write-down. The expected credit loss model will require a financial asset to be presented at the net amount expected to be collected. The allowance method for available-for-sale debt securities will allow the Company to record reversals of credit losses if the estimate of credit losses declines.

Income Tax Reform Update

On December 22, 2017, H.R.1, "An Act to Provide for Reconciliation Pursuant to Titles II and V of the Concurrent Resolution on the Budget for Fiscal Year 2018," was signed into law (Tax Reform Legislation).

Shortly after enactment, the Securities and Exchange Commission Staff issued Staff Accounting Bulletin No. 118 (SAB 118) to provide guidance on accounting for the Tax Reform Legislation impacts when the measurements of the income tax effects are complete, incomplete, or incomplete but for which a provisional amount can be estimated. SAB 118 permits the recognition of provisional amounts, and adjustments to provisional amounts, in subsequent reporting periods within the one year measurement period.

The Company has reflected the following incomplete but reasonably estimated provisional items in Deferred income taxes on the Condensed Consolidated Balance Sheet at March 31, 2018. The effects of the adjustments to the Company's provisional amounts for the three months ended March 31, 2018 did not impact income tax expense.

- The Company has recalculated its insurance reserves and the transition adjustment from existing law.
- The Company has recalculated amounts under special accounting method provisions for recognizing income for Federal income tax purposes no later than for financial accounting purposes and the transition adjustment from existing law.
- The Company has not recorded current or deferred taxes with respect to the international provisions since it does not expect to have inclusions in U.S. taxable income for certain earnings of foreign subsidiaries in future years.

Note B. Earnings (Loss) Per Share

Earnings (loss) per share is based on the weighted average number of outstanding common shares. Basic earnings (loss) per share excludes the impact of dilutive securities and is computed by dividing Net income (loss) by the weighted average number of common shares outstanding for the period. Diluted earnings (loss) per share reflects the potential dilution that could occur if securities or other contracts to issue common stock were exercised or converted into common stock.

For the three months ended March 31, 2018 and 2017, approximately 1,009 thousand and 974 thousand potential shares attributable to exercises or conversions into common stock under stock-based employee compensation plans were included in the calculation of diluted earnings per share. For those same periods, approximately 9 thousand and 148 thousand potential shares attributable to exercises or conversions into common stock under stock-based employee compensation plans were not included in the calculation of diluted earnings per share because the effect would have been antidilutive.

Note C . Investments

The significant components of Net investment income are presented in the following table.

Three months ended March 31

(In millions)	2018	2017
Fixed maturity securities	\$ 446	\$ 455
Equity securities	10	1
Limited partnership investments	30	90
Mortgage loans	11	7
Short term investments	6	3
Trading portfolio	2	2
Other		1
Gross investment income	505	559
Investment expense	(15)	(14)
Net investment income	\$ 490	\$ 545

During the three months ended March 31, 2018, less than \$1 million of Net investment income was recognized due to the change in fair value of common stock still held as of March 31, 2018.

Net realized investment gains (losses) are presented in the following table.

Three months ended March 31

(In millions)	2018	2017
Net realized investment gains (losses):		
Fixed maturity securities:		
Gross realized gains	\$ 69	\$ 49
Gross realized losses	 (51)	 (17)
Net realized investment gains (losses) on fixed maturity securities	18	32
Equity securities	(15)	_
Derivatives	5	1
Short term investments and other	 4	 3
Net realized investment gains (losses)	\$ 12	\$ 36

During the three months ended March 31, 2018, \$15 million of Net realized investment losses were recognized due to the change in fair value of non-redeemable preferred stock still held as of March 31, 2018.

The components of OTTI losses recognized in earnings by asset type are presented in the following table.

Three months ended March 31

(In millions)	2018		20	17
Fixed maturity securities available-for-sale:				
Corporate and other bonds	\$	5	\$	2
Asset-backed		1		_
OTTI losses recognized in earnings	\$	6	\$	2

The following tables present a summary of fixed maturity and equity securities.

March 31, 2018 (1) (In millions)	 Cost or Amortized Cost	Gross Unrealized Gains	Gross Unrealized Losses	Estimated Fair Value	Unrealized OTTI osses (Gains)
Fixed maturity securities available-for-sale:					
Corporate and other bonds	\$ 17,538	\$ 1,177	\$ 115	\$ 18,600	\$ _
States, municipalities and political subdivisions	11,682	1,205	12	12,875	_
Asset-backed:					
Residential mortgage-backed	5,050	83	78	5,055	(27)
Commercial mortgage-backed	1,948	29	22	1,955	_
Other asset-backed	 1,185	10	7	1,188	_
Total asset-backed	8,183	122	107	8,198	(27)
U.S. Treasury and obligations of government-sponsored enterprises	124	2	7	119	_
Foreign government	448	7	5	450	_
Redeemable preferred stock	 9	1	_	10	_
Total fixed maturity securities available-for-sale	37,984	2,514	246	40,252	\$ (27)
Total fixed maturity securities trading	7			7	
Total fixed maturity securities	\$ 37,991	\$ 2,514	\$ 246	\$ 40,259	

December 31, 2017 (In millions)	An	Cost or nortized Cost	Gross Unrealized Gains	Gross Unrealized Losses	Estimated Fair Value	I	Unrealized OTTI Losses (Gains)
Fixed maturity securities available-for-sale:							
Corporate and other bonds	\$	17,210	\$ 1,625	\$ 28	\$ 18,807	\$	
States, municipalities and political subdivisions		12,478	1,551	2	14,027		(11)
Asset-backed:							
Residential mortgage-backed		5,043	109	32	5,120		(27)
Commercial mortgage-backed		1,840	46	14	1,872		_
Other asset-backed		1,083	16	5	1,094		_
Total asset-backed		7,966	171	51	8,086		(27)
U.S. Treasury and obligations of government-sponsored enterprises		111	2	4	109		_
Foreign government		437	9	2	444		
Redeemable preferred stock		10	1	 	11		_
Total fixed maturity securities available-for-sale		38,212	3,359	87	41,484	\$	(38)
Total fixed maturity securities trading		3			3		
Equity securities available-for-sale:							
Common stock		21	7	1	27		
Preferred stock		638	31	1	668		
Total equity securities available-for-sale		659	38	2	695		
Total fixed maturity and equity securities	\$	38,874	\$ 3,397	\$ 89	\$ 42,182		

⁽¹⁾ As of January 1, 2018, the Company adopted ASU 2016-01, Financial Instruments - Overall (Subtopic 825-10): Recognition and Measurement of Financial Assets and Financial Liabilities. The change in fair value of equity securities is now recognized through the income statement. See Note A to the Condensed Consolidated Financial Statements for additional information.

The net unrealized gains on investments included in the tables above are recorded as a component of AOCI. When presented in AOCI, these amounts are net of tax and any required Shadow Adjustments. To the extent that unrealized gains on fixed income securities supporting certain products within the Life & Group segment would result in a premium deficiency if realized, a related increase in Insurance reserves is recorded, net of tax, as a reduction of net unrealized gains through Other comprehensive income (loss) (Shadow Adjustments). As of March 31, 2018 and December 31, 2017, the net unrealized gains on investments included in AOCI were correspondingly reduced by Shadow Adjustments of \$ 1,338 million and \$ 1,411 million.

The following tables present the estimated fair value and gross unrealized losses of fixed maturity and equity securities in a gross unrealized loss position by the length of time in which the securities have continuously been in that position.

		Less than	12 N	Ionths	12 Months or Longer						
March 31, 2018 ⁽¹⁾		Estimated		Gross Unrealized		Estimated	Gross Unrealized		Estimated		Gross Unrealized
(In millions)]	Fair Value		Losses	_	Fair Value	 Losses		Fair Value		Losses
Fixed maturity securities available-for-sale:											
Corporate and other bonds	\$	4,229	\$	104	\$	146	\$ 11	\$	4,375	\$	115
States, municipalities and political subdivisions		962		12		3	_		965		12
Asset-backed:											
Residential mortgage-backed		2,570		55		504	23		3,074		78
Commercial mortgage-backed		668		11		200	11		868		22
Other asset-backed		441		5		15	2		456		7
Total asset-backed		3,679		71		719	36		4,398		107
U.S. Treasury and obligations of government-sponsored enterprises		59		4		26	3		85		7
Foreign government		214		5		4			218		5
Total	\$	9,143	\$	196	\$	898	\$ 50	\$	10,041	\$	246

	Less tha	n 12 Months	12 Mont	hs or Longer	Total			
December 31, 2017 (In millions)	Estimated Fair Value	Gross Unrealized Losses	Estimated Fair Value	Gross Unrealized Losses	Estimated Fair Value	Gross Unrealized Losses		
Fixed maturity securities available-for-sale:								
Corporate and other bonds	\$ 1,354	\$ 21	\$ 168	\$ 7	\$ 1,522	\$ 28		
States, municipalities and political subdivisions	72	1	85	1	157	2		
Asset-backed:								
Residential mortgage-backed	1,228	5	947	27	2,175	32		
Commercial mortgage-backed	403	4	212	10	615	14		
Other asset-backed	248	3	18	2	266	5		
Total asset-backed	1,879	12	1,177	39	3,056	51		
U.S. Treasury and obligations of government-sponsored enterprises	49	2	21	2	70	4		
Foreign government	166	2	4		170	2		
Total fixed maturity securities available-for-sale	3,520	38	1,455	49	4,975	87		
Equity securities available-for-sale:								
Common stock	7	1	_	_	7	1		
Preferred stock	93	1			93	1		
Total equity securities available-for-sale	100	2	_		100	2		
Total	\$ 3,620	\$ 40	\$ 1,455	\$ 49	\$ 5,075	\$ 89		

⁽¹⁾ As of January 1, 2018, the Company adopted ASU 2016-01, Financial Instruments - Overall (Subtopic 825-10): Recognition and Measurement of Financial Assets and Financial Liabilities. The change in fair value of equity securities is now recognized through the income statement. See Note A to the Condensed Consolidated Financial Statements for additional information.

Based on current facts and circumstances, the Company believes the unrealized losses presented in the March 31, 2018 securities in a gross unrealized loss position table above are not indicative of the ultimate collectibility of the current amortized cost of the securities, but rather are attributable to changes in interest rates, credit spreads and other factors. The Company has no current intent to sell securities with unrealized losses, nor is it more likely than not that it will be required to sell prior to recovery of amortized cost; accordingly, the Company has determined that there are no additional OTTI losses to be recorded as of March 31, 2018.

The following table presents the activity related to the pretax credit loss component reflected in Retained earnings on fixed maturity securities still held as of March 31, 2018 and 2017 and for which a portion of an OTTI loss was recognized in Other comprehensive income (loss).

Three months ended March 31

(In millions)	2018		2017	
Beginning balance of credit losses on fixed maturity securities	\$	27	\$	36
Reductions for securities sold during the period		(2)		(4)
Ending balance of credit losses on fixed maturity securities	\$	25	\$	32

Contractual Maturity

The following table presents available-for-sale fixed maturity securities by contractual maturity.

		March	31, 20	018		Decembe	er 31,	r 31, 2017		
(In millions)	Cost or Amortized Cost		Estimated Fair Value		Cost or Amortized Cost			Estimated Fair Value		
Due in one year or less	\$	1,323	\$	1,343	\$	1,135	\$	1,157		
Due after one year through five years		8,277		8,495		8,165		8,501		
Due after five years through ten years		15,802		16,093		16,060		16,718		
Due after ten years		12,582		14,321		12,852		15,108		
Total	\$	37,984	\$	40,252	\$	38,212	\$	41,484		

Actual maturities may differ from contractual maturities because certain securities may be called or prepaid. Securities not due at a single date are allocated based on weighted average life.

Derivative Financial Instruments

The Company holds an embedded derivative on a funds withheld liability with a notional value of \$163 million and \$167 million as of March 31, 2018 and December 31, 2017 and a fair value of \$1 million and \$(3) million as of March 31, 2018 and December 31, 2017. The embedded derivative on the funds withheld liability is accounted for separately and reported with the funds withheld liability in Other liabilities on the Condensed Consolidated Balance Sheets.

Investment Commitments

As of March 31, 2018, the Company had committed approximately \$428 million to future capital calls from various third-party limited partnership investments in exchange for an ownership interest in the related partnerships.

As of March 31, 2018, the Company had mortgage loan commitments of \$47 million representing signed loan applications received and accepted.

The Company invests in various privately placed debt securities, including bank loans, as part of its overall investment strategy and has committed to additional future purchases, sales and funding. Purchases and sales of privately placed debt securities are recorded once funded. As of March 31, 2018, the Company had commitments to purchase or fund additional amounts of \$198 million and sell \$162 million under the terms of such securities.

Note D . Fair Value

Fair value is the price that would be received upon sale of an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. The following fair value hierarchy is used in selecting inputs, with the highest priority given to Level 1, as these are the most transparent or reliable.

Level 1 - Quoted prices for identical instruments in active markets.

Level 2 - Quoted prices for similar instruments in active markets; quoted prices for identical or similar instruments in markets that are not active; and model-derived valuations in which all significant inputs are observable in active markets.

Level 3 - Valuations derived from valuation techniques in which one or more significant inputs are not observable.

Prices may fall within Level 1, 2 or 3 depending upon the methodology and inputs used to estimate fair value for each specific security. In general, the Company seeks to price securities using third-party pricing services. Securities not priced by pricing services are submitted to independent brokers for valuation and, if those are not available, internally developed pricing models are used to value assets using a methodology and inputs the Company believes market participants would use to value the assets. Prices obtained from third-party pricing services or brokers are not adjusted by the Company.

The Company performs control procedures over information obtained from pricing services and brokers to ensure prices received represent a reasonable estimate of fair value and to confirm representations regarding whether inputs are observable or unobservable. Procedures may include i) the review of pricing service methodologies or broker pricing qualifications, ii) back-testing, where past fair value estimates are compared to actual transactions executed in the market on similar dates, iii) exception reporting, where period-over-period changes in price are reviewed and challenged with the pricing service or broker based on exception criteria, iv) deep dives, where the Company performs an independent analysis of the inputs and assumptions used to price individual securities and v) pricing validation, where prices received are compared to prices independently estimated by the Company.

Assets and Liabilities Measured at Fair Value

Assets and liabilities measured at fair value on a recurring basis are presented in the following tables. Corporate bonds and other includes obligations of the U.S. Treasury, government-sponsored enterprises and foreign governments and redeemable preferred stock.

March 31, 2018 (In millions)	I	Level 1	Level 2	Level 3	Total Assets/Liabilities at Fair Value
Assets					
Fixed maturity securities:					
Corporate bonds and other	\$	167	\$ 18,917	\$ 100	\$ 19,184
States, municipalities and political subdivisions		_	12,876	1	12,877
Asset-backed		_	7,919	279	8,198
Total fixed maturity securities		167	39,712	380	40,259
Equity securities:					
Common stock		83	_	4	87
Non-redeemable preferred stock		67	602	14	683
Total equity securities		150	602	18	770
Short term and other		156	978		1,134
Total assets	\$	473	\$ 41,292	\$ 398	\$ 42,163
Liabilities			 		
Other liabilities	\$	_	\$ (1)	\$ _	\$ (1)
Total liabilities	\$	_	\$ (1)	\$ _	\$ (1)
December 31, 2017					Total Assets/Liabilities
(In millions)	1	Level 1	 Level 2	 Level 3	 at Fair Value
Assets					
Fixed maturity securities:					
Corporate bonds and other	\$	128	\$ 19,148	\$ 98	\$ 19,374
States, municipalities and political subdivisions		_	14,026	1	14,027
Asset-backed			 7,751	 335	 8,086
Total fixed maturity securities		128	40,925	434	41,487
Equity securities:					
Common stock		23	_	4	27
Non-redeemable preferred stock		68	 584	16	 668
Total equity securities		91	584	20	695
Short term and other		396	958	 _	1,354
Total assets	\$	615	\$ 42,467	\$ 454	\$ 43,536
Liabilities					
Other liabilities	\$		\$ 3	\$ 	\$ 3

The tables below present a reconciliation for all assets and liabilities measured at fair value on a recurring basis using significant unobservable inputs (Level 3).

Level 3 (In millions)	Corporate bonds and other	States, municipalities and political subdivisions	Asset- backed	Equity securities	Derivative financial instruments	Total
Balance as of January 1, 2018	\$ 98	\$ 1	\$ 335	\$ 20	\$	\$ 454
Total realized and unrealized investment gains (losses):						
Reported in Net realized investment gains (losses)	(1)	_	7	(2)	_	4
Reported in Net investment income	_	_	_	_	_	_
Reported in Other comprehensive income			(5)			(5)
Total realized and unrealized investment gains (losses)	(1)	_	2	(2)	_	(1)
Purchases	_	_	30	_	_	30
Sales	_	_	(72)	_	_	(72)
Settlements	(2)	_	(6)	_	_	(8)
Transfers into Level 3	5	_	_	_	_	5
Transfers out of Level 3			(10)			(10)
Balance as of March 31, 2018	\$ 100	\$ 1	\$ 279	\$ 18	\$	\$ 398
Unrealized gains (losses) on Level 3 assets and liabilities held as of March 31, 2018 recognized in Net income (loss)	\$ <u> </u>	<u> </u>	<u> </u>	\$ (2)	<u> </u>	\$ (2)

Level 3 (In millions)	Corporate bonds and other	States, municipalities and political subdivisions	Asset-backed	Equity securities	Derivative financial instruments	Life settlement contracts	Total
Balance as of January 1, 2017	\$ 130	\$ 1	\$ 199	\$ 19	\$ —	\$ 58	\$ 407
Total realized and unrealized investment gains (losses):							
Reported in Net realized investment gains (losses)	_	_	_	_	1	_	1
Reported in Other revenues	_	_	_	_	_	6	6
Reported in Other comprehensive income	1	_	2	1	_	_	4
Total realized and unrealized investment gains (losses)	1	_	2	1	1	6	11
Purchases	5	_	38	1	_	_	44
Sales	(1)	_	_	(2)	(1)	(13)	(17)
Settlements	(14)	_	(6)	_	_	(5)	(25)
Transfers into Level 3	_	_	28	_	_	_	28
Transfers out of Level 3	_	_	(5)	_	_	_	(5)
Balance as of March 31, 2017	\$ 121	\$ 1	\$ 256	\$ 19	\$ —	\$ 46	\$ 443
Unrealized gains (losses) on Level 3 assets and liabilities held as of March 31, 2017 recognized in Net income (loss)	s –	\$ —	\$ —	s –	\$ —	s –	\$ —

Securities may be transferred in or out of levels within the fair value hierarchy based on the availability of observable market information and quoted prices used to determine the fair value of the security. The availability of observable market information and quoted prices varies based on market conditions and trading volume. During the three months ended March 31, 2018 there were \$29 million of transfers from Level 2 to Level 1 and no transfers from Level 1 to Level 2. During the three months ended March 31, 2017 there were no transfers between Level 1 and Level 2. The Company's policy is to recognize transfers between levels at the beginning of quarterly reporting periods.

Valuation Methodologies and Inputs

The following section describes the valuation methodologies and relevant inputs used to measure different financial instruments at fair value, including an indication of the level in the fair value hierarchy in which the instruments are generally classified.

Fixed Maturity Securities

Level 1 securities include highly liquid and exchange traded bonds and redeemable preferred stock, valued using quoted market prices. Level 2 securities include most other fixed maturity securities as the significant inputs are observable in the marketplace. All classes of Level 2 fixed maturity securities are valued using a methodology based on information generated by market transactions involving identical or comparable assets, a discounted cash flow methodology, or a combination of both when necessary. Common inputs for all classes of fixed maturity securities include prices from recently executed transactions of similar securities, marketplace quotes, benchmark yields, spreads off benchmark yields, interest rates and U.S. Treasury or swap curves. Specifically for asset-backed securities, key inputs include prepayment and default projections based on past performance of the underlying collateral and current market data. Fixed maturity securities are primarily assigned to Level 3 in cases where broker/dealer quotes are significant inputs to the valuation and there is a lack of transparency as to whether these quotes are based on information that is observable in the marketplace. Level 3 securities also include private placement debt securities whose fair value is determined using internal models with inputs that are not market observable.

Equity Securities

Level 1 securities include publicly traded securities valued using quoted market prices. Level 2 securities are primarily valued using pricing for similar securities, recently executed transactions and other pricing models utilizing market observable inputs. Level 3 securities are primarily priced using broker/dealer quotes and internal models with inputs that are not market observable.

Short Term and Other Invested Assets

Securities that are actively traded or have quoted prices are classified as Level 1. These securities include money market funds and treasury bills. Level 2 primarily includes commercial paper, for which all inputs are market observable. Fixed maturity securities purchased within one year of maturity are classified consistent with fixed maturity securities discussed above. Short term investments as presented in the tables above differ from the amounts presented on the Condensed Consolidated Balance Sheets because certain short term investments, such as time deposits, are not measured at fair value.

As of March 31, 2018 and December 31, 2017, there were approximately \$ 42 million and \$39 million of overseas deposits within other invested assets, which can be redeemed at net asset value in 90 days or less. Overseas deposits are excluded from the fair value hierarchy because their fair value is recorded using the net asset value per share (or equivalent) practical expedient.

Life Settlement Contracts

The Company sold its life settlement contracts to a third party in 2017. The valuation of the life settlement contracts was based on the terms of sale. The contracts were classified as Level 3 as there was not an active market for life settlement contracts.

Derivative Financial Investments

Level 2 investments primarily include the embedded derivative on the funds withheld liability. The embedded derivative on funds withheld liability is valued using the change in fair value of the assets supporting the funds withheld liability, which are fixed maturity securities valued with observable inputs.

Significant Unobservable Inputs

The following tables present quantitative information about the significant unobservable inputs utilized by the Company in the fair value measurements of Level 3 assets. Valuations for assets and liabilities not presented in the tables below are primarily based on broker/dealer quotes for which there is a lack of transparency as to inputs used to develop the valuations. The quantitative detail of these unobservable inputs is neither provided nor reasonably available to the Company.

March 31, 2018	Estimated Fair Value (In millions)	Valuation Technique(s)	Unobservable Input(s)	Range (Weighted Average)
Fixed maturity securities	\$ 133	Discounted cash flow	Credit spread	1% - 12% (3%)
Estimated Fair Value December 31, 2017 (In millions)		Valuation Technique(s)	Unobservable Input(s)	Range (Weighted Average)
Fixed maturity securities	\$ 136	Discounted cash flow	Credit spread	1% - 12% (3%)

For fixed maturity securities, an increase to the credit spread assumptions would result in a lower fair value measurement.

Financial Assets and Liabilities Not Measured at Fair Value

The carrying amount and estimated fair value of the Company's financial assets and liabilities which are not measured at fair value on the Condensed Consolidated Balance Sheets are presented in the following tables.

March 31, 2018		Carrying	Estimated Fair Value							
(In millions)		Amount			Level 1 Level 2			Level 3		Total
Assets										
Mortgage loans	\$	864	\$	_	\$	_	\$	856	\$	856
Note receivable		34		_		_		34		34
Liabilities										
Short term debt	\$	30	\$	_	\$	30	\$	_	\$	30
Long term debt		2,679		_		2,786		_		2,786
December 31, 2017		Carrying				Estimated	Fair	·Value		
(In millions)	_	Amount		Level 1	Level 2			Level 3	Total	
Assets										
Mortgage loans	\$	839	\$	_	\$	_	\$	844	\$	844
Note receivable		46		_		_		46		46
X 1 1 101.1										
Liabilities										
Short term debt	\$	150	\$	_	\$	150	\$	_	\$	150

The following methods and assumptions were used to estimate the fair value of these financial assets and liabilities.

The fair values of mortgage loans were based on the present value of the expected future cash flows discounted at the current interest rate for origination of similar quality loans, adjusted for specific loan risk.

The fair value of the note receivable was based on the present value of the expected future cash flows discounted at the current interest rate for origination of similar notes, adjusted for specific credit risk.

The Company's senior notes and debentures were valued based on observable market prices. The fair value for other debt was estimated using discounted cash flows based on current incremental borrowing rates for similar borrowing arrangements.

The carrying amounts reported on the Condensed Consolidated Balance Sheets for Cash, Short term investments not carried at fair value, Accrued investment income and certain Other assets and Other liabilities approximate fair value due to the short term nature of these items. These assets and liabilities are not listed in the tables above.

Note E. Claim and Claim Adjustment Expense Reserves

The Company's property and casualty insurance claim and claim adjustment expense reserves represent the estimated amounts necessary to resolve all outstanding claims, including incurred but not reported (IBNR) claims as of the reporting date. The Company's reserve projections are based primarily on detailed analysis of the facts in each case, the Company's experience with similar cases and various historical development patterns. Consideration is given to such historical patterns as claim reserving trends and settlement practices, loss payments, pending levels of unpaid claims and product mix, as well as court decisions, economic conditions, including inflation, and public attitudes. All of these factors can affect the estimation of claim and claim adjustment expense reserves.

Establishing claim and claim adjustment expense reserves, including claim and claim adjustment expense reserves for catastrophic events that have occurred, is an estimation process. Many factors can ultimately affect the final settlement of a claim and, therefore, the necessary reserve. Changes in the law, results of litigation, medical costs, the cost of repair materials and labor rates can affect ultimate claim costs. In addition, time can be a critical part of reserving determinations since the longer the span between the incidence of a loss and the payment or settlement of the claim, the more variable the ultimate settlement amount can be. Accordingly, short-tail claims, such as property damage claims, tend to be more reasonably estimable than long-tail claims, such as workers' compensation, general liability and professional liability claims. Adjustments to prior year reserve estimates, if necessary, are reflected in the results of operations in the period that the need for such adjustments is determined. There can be no assurance that the Company's ultimate cost for insurance losses will not exceed current estimates.

Catastrophes are an inherent risk of the property and casualty insurance business and have contributed to material period-to-period fluctuations in the Company's results of operations and/or equity. The Company reported catastrophe losses, net of reinsurance, of \$34 million for the three months ended March 31, 2018 and 2017. Net catastrophe losses in the first quarter of 2018 and 2017 related primarily to U.S. weather-related events.

Liability for Unpaid Claim and Claim Adjustment Expenses Rollforward

The following table presents a reconciliation between beginning and ending claim and claim adjustment expense reserves, including claim and claim adjustment expense reserves of the Life & Group segment.

For the three months ended March 31

(In millions)	2018		2017
Reserves, beginning of year:			
Gross	\$	22,004	\$ 22,343
Ceded		3,934	 4,094
Net reserves, beginning of year		18,070	 18,249
Net incurred claim and claim adjustment expenses:			
Provision for insured events of current year		1,246	1,207
Decrease in provision for insured events of prior years		(34)	(82)
Amortization of discount		47	 48
Total net incurred (1)		1,259	1,173
Net payments attributable to:			
Current year events		(91)	(68)
Prior year events		(1,219)	 (1,184)
Total net payments		(1,310)	(1,252)
Foreign currency translation adjustment and other		(9)	 14
Net reserves, end of period		18,010	18,184
Ceded reserves, end of period		4,057	 4,076
Gross reserves, end of period	\$	22,067	\$ 22,260

⁽¹⁾ Total net incurred above does not agree to Insurance claims and policyholders' benefits as reflected on the Condensed Consolidated Statements of Operations due to amounts related to retroactive reinsurance deferred gain accounting, uncollectible reinsurance and loss deductible receivables, and benefit expenses related to future policy benefits, which are not reflected in the table above.

Net Prior Year Development

Changes in estimates of claim and claim adjustment expense reserves, net of reinsurance, for prior years are defined as net prior year loss reserve development (development). These changes can be favorable or unfavorable. The following table presents development recorded for the Specialty, Commercial, International and Corporate & Other segments.

For the three months ended March 31

(In millions)	 2018	2017
Pretax (favorable) unfavorable development:		
Specialty	\$ (30)	\$ (12)
Commercial	(9)	(43)
International	_	(2)
Corporate & Other	 	_
Total pretax (favorable) unfavorable development	\$ (39)	\$ (57)

Specialty

The following table presents further detail of the development recorded for the Specialty segment.

Three months ended March 31

(In millions)	2018		2017
Pretax (favorable) unfavorable development:			
Medical Professional Liability	\$ 20) \$	20
Other Professional Liability and Management Liability	(34	4)	(32)
Surety	(1:	5)	_
Warranty	_	-	_
Other	(i)	
Total pretax (favorable) unfavorable development	\$ (30)) \$	(12)

<u>2018</u>

Unfavorable development for medical professional liability was primarily due to higher than expected severity in accident years 2014 and 2017 in our hospitals business.

Favorable development in other professional liability and management liability was primarily due to lower than expected claim frequency in accident years 2013 through 2015 related to financial institutions.

Favorable development for surety was due to lower than expected loss emergence for accident years 2015 and prior.

2017

Favorable development in other professional liability and management liability was primarily due to favorable settlements on closed claims and lower than expected frequency of large losses related to professional liability in accident years 2011 through 2016.

Unfavorable development in medical professional liability was primarily due to continued higher than expected frequency in aging services.

Commercial

The following table presents further detail of the development recorded for the Commercial segment.

Three months ended March 31

(In millions)	 2018		2017
Pretax (favorable) unfavorable development:			
Commercial Auto	\$ (1)	\$	(26)
General Liability	(8)		(18)
Workers' Compensation	(6)		
Property and Other	 6		1
Total pretax (favorable) unfavorable development	\$ (9)	\$	(43)

<u>2018</u>

Favorable development for general liability was primarily due to lower than expected frequency and severity in accident years 2015 and prior for our middle market construction business.

2017

Favorable development for commercial auto was primarily due to lower than expected severity in accident years 2013 through 2015.

Favorable development for general liability was due to lower than expected severity in life sciences.

International

The following table presents further detail of the development recorded for the International segment.

Three months ended March 31

(In millions)	2018	2017
Pretax (favorable) unfavorable development:		
Medical Professional Liability	\$ 1	\$
Other Professional Liability	1	(1)
Liability	_	_
Property & Marine	(2)	1
Other	_	(2)
Total pretax (favorable) unfavorable development	<u> </u>	\$ (2)

Asbestos and Environmental Pollution (A&EP) Reserves

In 2010, Continental Casualty Company (CCC) together with several of the Company's other insurance subsidiaries completed a transaction with National Indemnity Company (NICO), a subsidiary of Berkshire Hathaway Inc., under which substantially all of the Company's legacy A&EP liabilities were ceded to NICO through a Loss Portfolio Transfer (LPT). At the effective date of the transaction, the Company ceded approximately \$1.6 billion of net A&EP claim and allocated claim adjustment expense reserves to NICO under a retroactive reinsurance agreement with an aggregate limit of \$4 billion . The \$1.6 billion of claim and allocated claim adjustment expense reserves ceded to NICO was net of \$1.2 billion of ceded claim and allocated claim adjustment expense reserves under existing third-party reinsurance contracts. The NICO LPT aggregate reinsurance limit also covers credit risk on the existing third-party reinsurance related to these liabilities. The Company paid NICO a reinsurance premium of \$2 billion and transferred to NICO billed third-party reinsurance receivables related to A&EP claims with a net book value of \$215 million, resulting in total consideration of \$2.2 billion.

Subsequent to the effective date of the LPT, the Company recognized adverse prior year development on its A&EP reserves which resulted in additional amounts ceded under the LPT. As a result, the cumulative amounts ceded under the LPT exceeded the \$2.2 billion consideration paid, resulting in the NICO LPT moving into a gain position, requiring retroactive reinsurance accounting. Under retroactive reinsurance accounting, this gain is deferred and only recognized in earnings in proportion to actual paid recoveries under the LPT. Over the life of the contract, there is no economic impact as long as any additional losses incurred are within the limit of the LPT. In a period in which the Company recognizes a change in the estimate of A&EP reserves that increases the amounts ceded under the LPT, the proportion of actual paid recoveries to total ceded losses is affected and the change in the deferred gain is recognized in earnings as if the revised estimate of ceded losses was available at the effective date of the LPT. The effect of the deferred retroactive reinsurance benefit is recorded in Insurance claims and policyholders' benefits in the Condensed Consolidated Statement of Operations.

The following table presents the impact of the Loss Portfolio Transfer on the Condensed Consolidated Statements of Operations.

2	2018	2017			
\$	113	\$	60		
	(16)		_		
	97		60		
	(57)		(40)		
\$	40	\$	20		
		(16) 97 (57)	\$ 113 \$ (16) 97 (57)		

Based upon the Company's annual A&EP reserve review, net unfavorable prior year development of \$113 million and \$60 million was recognized before consideration of cessions to the LPT for the three months ended March 31, 2018 and 2017. Additionally, in 2018, the Company released a portion of its provision for uncollectible third party reinsurance. The 2018 unfavorable development was driven by higher than anticipated defense costs on direct asbestos environmental accounts and paid losses on assumed reinsurance exposures. The 2017 unfavorable development was driven by modestly higher anticipated payouts on claims from known sources of asbestos exposure. While the unfavorable development was ceded to NICO under the LPT, the Company's Net income in the periods was negatively affected due to the application of retroactive reinsurance accounting.

As of March 31, 2018 and December 31, 2017, the cumulative amounts ceded under the LPT were \$3.0 billion and \$2.9 billion. The unrecognized deferred retroactive reinsurance benefit was \$366 million and \$326 million as of March 31, 2018 and December 31, 2017.

NICO established a collateral trust account as security for its obligations to the Company. The fair value of the collateral trust account was \$2.9 billion and \$3.1 billion as of March 31, 2018 and December 31, 2017. In addition, Berkshire Hathaway Inc. guaranteed the payment obligations of NICO up to the aggregate reinsurance limit as well as certain of NICO's performance obligations under the trust agreement. NICO is responsible for claims handling and collection from third-party reinsurers related to the Company's A&EP claims.

Note F. Legal Proceedings, Contingencies and Guarantees

CNA 401(k) Plus Plan Litigation

In September 2016, a class action lawsuit was filed against CCC, Continental Assurance Company (CAC) (a former subsidiary of CCC), CNAF, the Investment Committee of the CNA 401(k) Plus Plan (Plan), The Northern Trust Company and John Does 1-10 (collectively Defendants) related to the Plan. The complaint alleges that Defendants breached fiduciary duties to the Plan and caused prohibited transactions in violation of the Employee Retirement Income Security Act of 1974 when the Plan's Fixed Income Fund's annuity contract with CAC was canceled. The plaintiff alleges he and a proposed class of Plan participants who had invested in the Fixed Income Fund suffered lower returns in their Plan investments as a consequence of these alleged violations and seeks relief on behalf of the putative class. The Plan trustees have provided notice to their fiduciary coverage insurance carriers.

The plaintiff, Defendants and the Plan's fiduciary insurance carriers have agreed on terms to settle this matter. Upon execution of final settlement agreements, plaintiff and Defendants will propose a class settlement for court approval. Based on the executed term sheet, management has recorded its best estimate of the Company's probable loss and does not believe that the ultimate resolution of this matter will have a material impact on the Company's results of operations or financial position.

Small Business Premium Rate Adjustment

In 2016 and 2017, the Company identified rating errors related to its multi-peril package product and workers' compensation policies within its Small Business unit and determined that it would voluntarily issue premium refunds along with interest on affected policies. After the rating errors were identified, written and earned premium have been reported net of any impact from the premium rate adjustments. In the first quarter of 2017, the Company recorded a charge which reduced Earned premium by \$38 million and increased Interest expense by \$5 million for interest due to policyholders on the premium rate adjustments.

The policyholder refunds for the multi-peril package product were issued in the third quarter of 2017. The policyholder refunds for workers' compensation policies are expected to be refunded in the second half of 2018 and, as such, an additional \$1 million of Interest expense was recorded in the first quarter of 2018. The estimated refund liability, including interest, for the workers' compensation policies as of March 31, 2018 was \$60 million.

Other Litigation

The Company is a party to other routine litigation incidental to its business, which, based on the facts and circumstances currently known, is not material to the Company's results of operations or financial position.

Guarantees

As of March 31, 2018 and December 31, 2017, the Company had recorded liabilities of approximately \$5 million related to guarantee and indemnification agreements and management believes that it is not likely that any future indemnity claims will be significantly greater than the amounts recorded.

In the course of selling business entities and assets to third parties, the Company agreed to guarantee the performance of certain obligations of previously owned subsidiaries and to indemnify purchasers for losses arising out of breaches of representations and warranties with respect to the business entities or assets sold, including, in certain cases, losses arising from undisclosed liabilities or certain named litigation. Such guarantee and indemnification agreements in effect for sales of business entities, assets and third-party loans may include provisions that survive indefinitely. As of March 31, 2018, the aggregate amount related to quantifiable guarantees was \$375 million and the aggregate amount related to quantifiable indemnification agreements was \$252 million. In certain cases, should the Company be required to make payments under any such guarantee, it would have the right to seek reimbursement from an affiliate of a previously owned subsidiary.

In addition, the Company has agreed to provide indemnification to third-party purchasers for certain losses associated with sold business entities or assets that are not limited by a contractual monetary amount. As of March 31, 2018 the Company had outstanding unlimited indemnifications in connection with the sales of certain of its business entities or assets that included tax liabilities arising prior to a purchaser's ownership of an entity or asset, defects in title at the time of sale, employee claims arising prior to closing and in some cases losses arising from certain litigation and undisclosed liabilities. Certain provisions of the indemnification agreements survive

indefinitely, while others survive until the applicable statutes of limitation expire, or until the agreed-upon contract terms expire.

The Company also provided guarantees, if the primary obligor fails to perform, to holders of structured settlement annuities provided by a previously owned subsidiary. As of March 31, 2018, the potential amount of future payments the Company could be required to pay under these guarantees was approximately \$1.8 billion, which will be paid over the lifetime of the annuitants. The Company does not believe any payment is likely under these guarantees, as the Company is the beneficiary of a trust that must be maintained at a level that approximates the discounted reserves for these annuities.

Note G . Benefit Plans

The components of net periodic pension cost (benefit) are presented in the following table.

Three months ended March 31	Pension Plans					
(In millions)	2018	2017				
Net periodic pension cost (benefit)						
Service cost	\$ —	\$ —				
Non-service cost:						
Interest cost on projected benefit obligation	23	26				
Expected return on plan assets	(40)	(39)				
Amortization of net actuarial loss	9	9				
Settlement loss	4	2				
Total non-service cost	(4)	(2)				
Total net periodic pension cost (benefit)	\$ (4)	\$ (2)				

For the three months ended March 31, 2018, the Company recognized \$1 million of non-service cost in Insurance claims and policyholders' benefits and \$3 million of non-service cost in Other operating expenses. For the three months ended March 31, 2017, the Company recognized \$1 million of non-service costs in Insurance claims and policyholders' benefits and \$1 million of non-service costs in Other operating expenses.

Note H . Accumulated Other Comprehensive Income (Loss) by Component

The tables below display the changes in Accumulated other comprehensive income (loss) by component.

(In millions)	(losses	Net unrealized gains osses) on investments with OTTI losses		unrealized gains osses) on other investments	Pension and postretirement benefits			umulative foreign rrency translation adjustment	Total
Balance as of January 1, 2018, as previously reported	\$	25	\$	750	\$	(645)	\$	(98)	\$ 32
Cumulative effect adjustment from accounting change for adoption of ASU 2018-02 (1)		5		137		(130)		_	12
Cumulative effect adjustment from accounting change for adoption of ASU 2016-01 ⁽¹⁾ net of tax (expense) benefit of \$-, \$8, \$-, \$- and \$8				(28)				_	(28)
Balance as of January 1, 2018, as adjusted		30		859		(775)		(98)	16
Other comprehensive income (loss) before reclassifications		(10)		(414)		_		12	(412)
Amounts reclassified from accumulated other comprehensive income (loss) net of tax (expense) benefit of \$-, \$(4), \$3, \$- and \$(1)		(1)		15		(10)		_	4
Other comprehensive income (loss) net of tax (expense) benefit of \$2, \$109, \$(3), \$- and \$108		(9)		(429)		10		12	(416)
Balance as of March 31, 2018	\$	21	\$	430	\$	(765)	\$	(86)	\$ (400)

(1) See Note A to the Condensed Consolidated Financial Statements for additional information.

(In millions)	(losses)	realized gains on investments OTTI losses	Net unrealized gains (losses) on other investments			Pension and postretirement benefits	umulative foreign rrency translation adjustment	Total
Balance as of January 1, 2017	\$	30	\$	642	\$	(647)	\$ (198)	\$ (173)
Other comprehensive income (loss) before reclassifications		_		85		_	11	96
Amounts reclassified from accumulated other comprehensivincome (loss) net of tax (expense) benefit of \$(2), \$(9), \$4, \$ and \$(7)		4		18		(7)	_	15
Other comprehensive income (loss) net of tax (expense) benefit of \$1, \$(38), \$(4), \$- and \$(41)		(4)		67		7	11	81
Balance as of March 31, 2017	\$	26	\$	709	\$	(640)	\$ (187)	\$ (92)

Amounts reclassified from Accumulated other comprehensive income (loss) shown above are reported in Net income (loss) as follows:

Net realized investment gains (losses)
Net realized investment gains (losses)
Other operating expenses and Insurance claims and policyholders' benefits

Note I. Business Segments

The Company's property and casualty commercial insurance operations are managed and reported in three business segments: Specialty, Commercial and International. These three segments are collectively referred to as Property & Casualty Operations. The Company's operations outside of Property & Casualty Operations are managed and reported in two segments: Life & Group and Corporate & Other.

Effective January 1, 2018, management changed the segment presentation of the life sciences business and technology and media related errors and omissions business within the Specialty and Commercial business segments. The life sciences business moved from the Specialty business segment to the Commercial business segment and the technology and media related errors and omissions business moved from the Commercial business segment to the Specialty business segment. The new management responsibility for these businesses better aligns with line of business underwriting expertise and the manner in which the products are sold. Prior period information has been conformed to the new segment presentation.

The accounting policies of the segments are the same as those described in Note A to the Consolidated Financial Statements within CNAF's Annual Report on Form 10-K for the year ended December 31, 2017. The Company manages most of its assets on a legal entity basis, while segment operations are generally conducted across legal entities. As such, only Insurance and Reinsurance receivables, Insurance reserves, Deferred acquisition costs and Goodwill are readily identifiable for individual segments. Distinct investment portfolios are not maintained for every individual segment; accordingly, allocation of assets to each segment is not performed. Therefore, a significant portion of Net investment income and Realized investment gains or losses are allocated primarily based on each segment's net carried insurance reserves, as adjusted. All significant intersegment income and expense has been eliminated. Income taxes have been allocated on the basis of the taxable income of the segments.

In the following tables, certain financial measures are presented to provide information used by management to monitor the Company's operating performance. Management utilizes these financial measures to monitor the Company's insurance operations and investment portfolio.

Core income (loss), which is derived from certain income statement amounts, is used by management to monitor performance of the Company's insurance operations. The Company's investment portfolio is monitored by management through analysis of various factors including unrealized gains and losses on securities, portfolio duration and exposure to market and credit risk.

Core income (loss) is calculated by excluding from net income (loss) the after-tax effects of i) net realized investment gains (losses), ii) income or loss from discontinued operations, iii) any cumulative effects of changes in accounting guidance and iv) deferred tax asset and liability remeasurement as a result of an enacted U.S. Federal tax rate change. The calculation of core income (loss) excludes net realized investment gains or losses because net realized investment gains or losses are generally driven by economic factors that are not necessarily consistent with key drivers of underwriting performance, and are therefore not considered an indication of trends in insurance operations.

The Company's results of operations and selected balance sheet items by segment are presented in the following tables.

Three months ended March 31, 2018 (In millions)	Sı	oecialty	Com	mercial	Int	ternational		Life & Group		orpo & Otl		Elim	inations		Total
Operating revenues															
Net earned premiums	\$	672	\$	743	\$	236	\$	134	\$		_	\$	_	\$	1,785
Net investment income		122		149		14		200			5		_		490
Non-insurance warranty revenue		238		_		_		_			_		_		238
Other revenues		1		8		_		1			1		(1)		10
Total operating revenues		1,033		900		250		335			6		(1)		2,523
Claims, benefits and expenses															
Net incurred claims and benefits		379		468		142		303			41		_		1,333
Policyholders' dividends		1		5		_		_			_		_		6
Amortization of deferred acquisition costs		145		121		30		_			_		_		296
Non-insurance warranty expense		216		_		_		_			_		_		216
Other insurance related expenses		64		127		56		30			_		_		277
Other expenses		11		11		(4)		2			42		(1)		61
Total claims, benefits and expenses		816		732		224		335			83		(1)		2,189
Core income (loss) before income tax		217		168		26		_			(77)		_		334
Income tax (expense) benefit on core income (loss)		(46)		(35)		(3)		14			17		_		(53)
Core income (loss)	\$	171	\$	133	\$	23	\$	14	\$		(60)	\$	_		281
Net realized investment gains (losses)														='	12
Income tax (expense) benefit on net realized investment gains (losses)															(2)
Net realized investment gains (losses), after tax															10
Net income														\$	291
March 31, 2018															
(In millions)															
Reinsurance receivables		\$	737	\$	690	\$	244	\$	436	\$	2,330	\$	_	\$	4,437
Insurance receivables			947		1,140		315		11		1		_		2,414
Deferred acquisition costs			318		241		106		_		_		_		665
Goodwill			117		_		32		_		_		_		149
Insurance reserves															
Claim and claim adjustment expenses			5,696		8,718	1	,718	3	,501		2,434		_		22,067
Unearned premiums			2,063		1,490		559		145		_		(1)		4,256
Future policy benefits			_		_		_	10	,783		_		_		10,783
			2	_											

Three months ended March 31, 2017								Life &		Corporate				
(In millions)	Sp	ecialty	Com	mercial	Int	ternation	al	Group		& O		Elimi	nations	 Total
Operating revenues														
Net earned premiums	\$	654	\$	661	\$	197	7 \$	133	\$		_	\$	_	\$ 1,645
Net investment income		148		183		12	2	197			5		_	545
Non-insurance warranty revenue		93		_		_	-	_			_		_	93
Other revenues		1		9		_	<u> </u>	1					_	 11
Total operating revenues		896		853	_	209		331			5			 2,294
Claims, benefits and expenses														
Net incurred claims and benefits		400		423		115	5	330			21		_	1,289
Policyholders' dividends		1		3		_	-	_			_		_	4
Amortization of deferred acquisition costs		142		117		40	5	_			_		_	305
Non-insurance warranty expense		70		_		_	-	_			_		_	70
Other insurance related expenses		67		128		27	7	32			_		_	254
Other expenses		10		15		(6	5)	2			44		_	 65
Total claims, benefits and expenses		690		686		182	2	364			65		_	 1,987
Core income (loss) before income tax		206		167		27	7	(33)		(60)		_	307
Income tax (expense) benefit on core income (loss)		(69)		(56)		(7)	37			23		_	 (72)
Core income (loss)	\$	137	\$	111	\$	20	\$	4	\$		(37)	\$		 235
Net realized investment gains (losses)														36
Income tax (expense) benefit on net realized investment gains (losses)														 (11)
Net realized investment gains (losses), after tax														25
Net income														\$ 260
December 21, 2017														
December 31, 2017 (In millions)														
Reinsurance receivables		\$	671	\$	654	\$	212	\$	438	\$	2,315	\$		\$ 4,290
Insurance receivables		Þ	969		1,103	Ф	254	Ф	438	Ф	2,313		_	\$ 2,336
			318		223		93		0		2		_	634
Deferred acquisition costs Goodwill			117				31		_		_		_	148
Insurance reserves			117		_		31		_		_		_	140
Claim and claim adjustment expenses			5,669		8,764		1,636		3,499		2,436			22,004
Unearned premiums			2,020		1,409		472		128		2,730			4,029
Future policy benefits			_,020					1	1,179		_		_	11,179
r y /								•	,					-,

The following table presents operating revenue by line of business for each reportable segment.

Three months ended March 31

(In millions)	2018	2017
Specialty		
Management & Professional Liability	\$ 624	\$ 638
Surety	129	124
Warranty & Alternative Risks (1)	280	134
Specialty revenues	1,033	896
Commercial		
Middle Market	504	468
Small Business	119	96
Other Commercial Insurance	277	289
Commercial revenues	900	853
International		
Canada	58	51
CNA Europe	88	72
Hardy	104	86
International revenues	250	209
Life & Group revenues	335	331
Corporate & Other revenues	6	5
Eliminations	(1)	
Total operating revenues	2,523	2,294
Net realized investment gains (losses)	12	36
Total revenues	\$ 2,535	\$ 2,330

⁽¹⁾ As of January 1, 2018, the Company adopted ASU 2014-09 Revenue Recognition (Topic 606): Revenue from Contracts with Customers . See Note A to the Condensed Consolidated Financial Statements for additional information.

Note J. Non-Insurance Revenues from Contracts with Customers

Non-Insurance revenue is recognized when obligations under the terms of a contract with a customer are satisfied; generally this occurs over time as obligations are fulfilled. Revenue is measured as the amount of consideration the Company expects to receive in exchange for providing services.

Deferred Non-Insurance Warranty Revenue

Non-insurance warranty revenue is primarily generated from separately-priced service contracts that provide mechanical breakdown and other coverages to vehicle or consumer goods owners. The warranty contracts generally provide coverage from 1 month to 10 years. For warranty products in which the Company acts as the principal in the transaction, Non-insurance warranty revenues are reported on a gross basis, with amounts billed to customers reported as Non-insurance warranty revenue and commissions paid to agents reported as Non-insurance warranty expense. Non-insurance warranty revenue is reported net of any premiums related to contractual liability coverage issued by the Company's insurance operations. Additionally, the Company provides warranty administration services for dealer and manufacturer obligor warranty products, which include limited warranties and guaranteed automobile protection waivers.

The Company recognizes Non-insurance warranty revenues over the service period in proportion to the actuarially determined expected claims emergence pattern. Customers pay in full at the inception of the warranty contract. A liability for deferred revenue is recorded when cash payments are received or due in advance of the Company's performance. The deferred revenue balance includes amounts which are refundable on a pro rata basis upon cancellation.

The Company had deferred non-insurance warranty revenue balances of \$2.9 billion and \$3.0 billion reported in Other liabilities as of January 1, 2018 and March 31, 2018. The increase in the deferred revenue balance for the three months ended March 31, 2018 was primarily driven by cash payments received or due in advance of satisfying the Company's performance obligations, offset by cancellations and revenues recognized during the period. For the three months ended March 31, 2018, the Company recognized \$222 million of revenues that were included in the deferred revenue balance at the beginning of the period. For the three months ended March 31, 2018, Non-insurance warranty revenue recognized from performance obligations related to prior periods due to a change in estimate was not material. The Company expects to recognize approximately \$659 million of the deferred revenue in the remainder of 2018, \$761 million in 2019, \$593 million in 2020, and \$1 billion thereafter.

Cost to Obtain and Fulfill Non-Insurance Warranty Contracts with Customers

Dealers, retailers and agents earn commission for assisting the Company in obtaining non-insurance warranty contracts. Additionally, the Company utilizes a third-party to perform warranty administrator services for its consumer good warranties. These costs are deferred and recorded as Other assets. These costs are amortized to Non-insurance warranty expense consistent with how the related revenue is recognized.

A premium deficiency arises to the extent that estimated future costs associated with these contracts exceed unrecognized revenue. The Company evaluates deferred costs for recoverability as part of our premium deficiency assessment. Anticipated investment income is considered in the determination of the recoverability of deferred costs. If necessary, adjustments to deferred costs and a premium deficiency reserve, if any, are recorded in current period results of operations. No premium deficiency was recognized in the three months ended March 31, 2018.

As of March 31, 2018, capitalized commission costs were \$2.2 billion and capitalized administrator service costs were \$17 million. For the three months ended March 31, 2018, the amount of amortization of capitalized costs was \$157 million and there was no impairment loss related to the costs capitalized.

Item 2. Management's Discussion and Analysis (MD&A) of Financial Condition and Results of Operations

OVERVIEW

The following discussion highlights significant factors affecting the Company. References to "we," "our," "us" or like terms refer to the business of CNA. Based on 2016 statutory net written premiums, we are the eighth largest commercial insurer in the United States of America.

The following discussion should be read in conjunction with the Condensed Consolidated Financial Statements included under Part I, Item 1 of this Form 10-Q and Item 1A Risk Factors and Item 7 Management's Discussion and Analysis of Financial Condition and Results of Operations, which are included in our Annual Report on Form 10-K filed with the Securities and Exchange Commission for the year ended December 31, 2017.

We utilize the core income (loss) financial measure to monitor our operations. Core income (loss) is calculated by excluding from net income (loss) the after-tax effects of i) net realized investment gains or losses, ii) income or loss from discontinued operations, iii) any cumulative effects of changes in accounting guidance and iv) deferred tax asset and liability remeasurement as a result of an enacted U.S. Federal tax rate change. The calculation of core income (loss) excludes net realized investment gains or losses because net realized investment gains or losses are generally driven by economic factors that are not necessarily consistent with key drivers of underwriting performance, and are therefore not considered an indication of trends in insurance operations. Management monitors core income (loss) for each business segment to assess segment performance. Presentation of consolidated core income (loss) is deemed to be a non-GAAP financial measure. See further discussion regarding how we manage our business in Note I to the Condensed Consolidated Financial Statements included under Part I, Item 1. For reconciliations of non-GAAP measures to the most comparable GAAP measures and other information, please refer herein and/or to CNA's most recent 10-K on file with the Securities and Exchange Commission.

In evaluating the results of our Specialty, Commercial and International segments, we utilize the loss ratio, the expense ratio, the dividend ratio and the combined ratio. These ratios are calculated using GAAP financial results. The loss ratio is the percentage of net incurred claim and claim adjustment expenses to net earned premiums. The expense ratio is the percentage of insurance underwriting and acquisition expenses, including the amortization of deferred acquisition cost, to net earned premiums. The dividend ratio is the ratio of policyholders' dividends incurred to net earned premiums. The combined ratio is the sum of the loss, expense and dividend ratios. In addition we also utilize renewal premium change, rate, retention and new business in evaluating operating trends. Renewal premium change represents the estimated change in average premium on policies that renew, including rate and exposure changes. Rate represents the average change in price on policies that renew excluding exposure change. Exposure represents the measure of risk used in the pricing of the insurance product. Retention represents the percentage of premium dollars renewed in comparison to the expiring premium dollars from policies available to renew. Renewal premium change, rate and retention presented for the prior year are updated to reflect subsequent activity on policies written in the period. New business represents premiums from policies written with new customers and additional policies written with existing customers.

Changes in estimates of claim and claim adjustment expense reserves, net of reinsurance, for prior years are defined as net prior year loss reserve development within this MD&A. These changes can be favorable or unfavorable. Net prior year loss reserve development does not include the effect of related acquisition expenses. Further information on our reserves is provided in Note E to the Condensed Consolidated Financial Statements included under Part I, Item 1.

CRITICAL ACCOUNTING ESTIMATES

The preparation of the Condensed Consolidated Financial Statements in conformity with GAAP requires us to make estimates and assumptions that affect the reported amounts of assets and liabilities and the disclosure of contingent assets and liabilities at the date of the Condensed Consolidated Financial Statements and the amount of revenues and expenses reported during the period. Actual results may differ from those estimates.

Our Condensed Consolidated Financial Statements and accompanying notes have been prepared in accordance with GAAP applied on a consistent basis. We continually evaluate the accounting policies and estimates used to prepare the Condensed Consolidated Financial Statements. In general, our estimates are based on historical experience, evaluation of current trends, information from third-party professionals and various other assumptions that are believed to be reasonable under the known facts and circumstances.

The accounting estimates below are considered by us to be critical to an understanding of our Condensed Consolidated Financial Statements as their application places the most significant demands on our judgment:

- Insurance Reserves
- · Reinsurance and Insurance Receivables
- Valuation of Investments and Impairment of Securities
- Long Term Care Policies
- Income Taxes

Due to the inherent uncertainties involved with these types of judgments, actual results could differ significantly from estimates and may have a material adverse impact on our results of operations, equity, business, and insurer financial strength and corporate debt ratings. See the Critical Accounting Estimates section of our Management's Discussion and Analysis of Financial Condition and Results of Operations included under Item 7 of our Annual Report on Form 10-K for the year ended December 31, 2017 for further information.

CONSOLIDATED OPERATIONS

The following table includes the consolidated results of our operations including our financial measure, Core income (loss). For more detailed components of our business operations and the core income (loss) financial measure, see the segment discussions within this MD&A. For further discussion of Net investment income and Net realized investment results, see the Investments section of this MD&A.

Three months ended March 31

(In millions)	2018	2017
Operating Revenues		
Net earned premiums	\$ 1,785	\$ 1,645
Net investment income	490	545
Non-insurance warranty revenue	238	93
Other revenues	10	11
Total operating revenues	2,523	2,294
Claims, Benefits and Expenses		
Net incurred claims and benefits	1,333	1,289
Policyholders' dividends	6	4
Amortization of deferred acquisition costs	296	305
Other insurance related expenses	277	254
Non-insurance warranty expense	216	70
Other expenses	61	65
Total claims, benefits and expenses	2,189	1,987
Core income before income tax	334	307
Income tax expense on core income	(53)	(72)
Core income	281	235
Net realized investment gains	12	36
Income tax expense on net realized investment gains	(2)	(11)
Net realized investment gains, after tax	10	25
Net income	\$ 291	\$ 260

Core income increased \$46 million for the three months ended March 31, 2018 as compared with 2017. Excluding the effect of the corporate tax rate change, core income for our Property & Casualty Operations increased approximately \$4 million. Improved underwriting results more than offset lower net investment income driven by lower limited partnership returns. Excluding the effect of the Federal corporate income tax rate change, core income for our Life & Group segment improved \$22 million while core loss for our Corporate & Other segment increased \$12 million.

Net catastrophe losses were \$ 34 million for the three months ended March 31, 2018 and 2017. Favorable net prior year loss reserve development of \$39 million and \$57 million was recorded in the three months ended March 31, 2018 and 2017 related to our Specialty, Commercial, International and Corporate & Other segments. Further information on net prior year loss reserve development is in Note E to the Condensed Consolidated Financial Statements included under Part I, Item 1.

SEGMENT RESULTS

The following discusses the results of operations for our business segments. Our property and casualty commercial insurance operations are managed and reported in three business segments: Specialty, Commercial and International, which we refer to collectively as Property & Casualty Operations. Our operations outside of Property & Casualty Operations are managed and reported in two segments: Life & Group and Corporate & Other.

Specialty

The following table details the results of operations for Specialty.

Three months ended March 31

(In millions, except ratios, rate, renewal premium change and retention)	2018		2017
Net written premiums	\$ 686	\$	670
Net earned premiums	672		654
Net investment income	122		148
Core income	171		137
Other performance metrics:			
Loss and loss adjustment expense ratio	56.3	%	61.2%
Expense ratio	31.0		31.9
Dividend ratio	0.2		0.1
Combined ratio	87.5	%	93.2%
Rate	2	%	1%
Renewal premium change	3		4
Retention	85		88
New business	\$ 81	\$	55

Net written premiums for Specialty increased \$16 million for the three months ended March 31, 2018 as compared with the same period in 2017 due to higher new business and positive renewal premium change. The increase in net earned premiums was consistent with the trend in net written premiums.

Core income increased \$34 million for the three months ended March 31, 2018 as compared with the same period in 2017. Excluding the effect of the Federal corporate income tax rate change, core income increased approximately \$6 million. Improved underwriting results more than offset lower net investment income driven by lower limited partnership returns.

The combined ratio improved 5.7 points for the three months ended March 31, 2018 as compared with the same period in 2017. The loss ratio improved 4.9 points primarily due to 2.7 points of improvement in the current accident year loss ratio and higher favorable net prior year loss reserve development. Net catastrophe losses were \$3 million, or 0.5 points of the loss ratio, for the three months ended March 31, 2018, as compared to \$4 million, or 0.6 points of the loss ratio, for the three months ended March 31, 2018 as compared with the same period in 2017.

Favorable net prior year loss reserve development of \$30 million and \$12 million was recorded for the three months ended March 31, 2018 and 2017. Further information on net prior year loss reserve development is in Note E to the Condensed Consolidated Financial Statements included under Part I, Item 1.

The following table summarizes the gross and net carried reserves for Specialty.

(In millions)	Marc	ch 31, 2018	Decem	ber 31, 2017
Gross case reserves	\$	1,734	\$	1,742
Gross IBNR reserves		3,962		3,927
Total gross carried claim and claim adjustment expense reserves	\$	5,696	\$	5,669
Net case reserves	\$	1,585	\$	1,600
Net IBNR reserves		3,385		3,407
Total net carried claim and claim adjustment expense reserves	\$	4,970	\$	5,007

Commercial

The following table details the results of operations for Commercial.

Three months ended March 31

(In millions, except ratios, rate, renewal premium change and retention)	2018	2017
Net written premiums	\$ 832 \$	724
Net earned premiums	743	661
Net investment income	149	183
Core income	133	111
Other performance metrics:		
Loss and loss adjustment expense ratio	63.0%	63.9%
Expense ratio	33.5	37.3
Dividend ratio	 0.6	0.5
Combined ratio	 97.1%	101.7%
Rate	1%	0%
Renewal premium change	3	4
Retention	84	85
New business	\$ 181 \$	140

Net written premiums for Commercial were \$108 million higher for the three months ended March 31, 2018 as compared with the same period in 2017. The prior period included an unfavorable premium rate adjustment in Small Business which affected both net written premiums and net earned premiums as more fully discussed in Note F to the Condensed Consolidated Financial Statements under Part 1, Item 1. Excluding the Small Business premium rate adjustment, net written premiums increased \$61 million driven by higher new business and positive renewal premium change. The increase in net earned premiums was consistent with the trend in net written premiums.

Core income increased \$22 million for the three months ended March 31, 2018 as compared with the same period in 2017. Excluding the effect of the Federal corporate income tax rate change and the Small Business premium rate adjustment in the prior period, core income decreased approximately \$26 million due to improved underwriting results more than offset by lower net investment income driven by lower limited partnership returns.

Excluding the impact of the Small Business premium rate adjustment, the combined ratio increased 0.9 points, driven by 5.0 points of less favorable net prior year loss reserve development largely offset by a 2.4 point improvement in the current accident year loss ratio and a 1.8 point decrease in the expense ratio due to higher net earned premiums. Net catastrophe losses were \$29 million, or 3.9 points of the loss ratio, for the three months ended March 31, 2018, as compared to \$27 million, or 3.9 points of the loss ratio, for the three months ended March 31, 2017.

Favorable net prior year loss reserve development of \$9 million and \$43 million was recorded for the three months ended March 31, 2018 and 2017. Further information on net prior year loss reserve development is in Note E to the Condensed Consolidated Financial Statements included under Part I, Item 1.

The following table summarizes the gross and net carried reserves for Commercial.

(In millions)	Mar	ch 31, 2018	Decem	ber 31, 2017
Gross case reserves	\$	4,327	\$	4,427
Gross IBNR reserves		4,391		4,337
Total gross carried claim and claim adjustment expense reserves	\$	8,718	\$	8,764
Net case reserves	\$	4,008	\$	4,103
Net IBNR reserves		4,058		4,033
Total net carried claim and claim adjustment expense reserves	\$	8,066	\$	8,136

International

The following table details the results of operations for International.

Three months ended March 31

(In millions, except ratios, rate, renewal premium change and retention)	 2018	2017
Net written premiums	\$ 295	\$ 238
Net earned premiums	236	197
Net investment income	14	12
Core income	23	20
Other performance metrics:		
Loss and loss adjustment expense ratio	60.4%	58.3%
Expense ratio	 36.2	36.8
Combined ratio	96.6%	95.1%
Rate	2%	1%
Renewal premium change	8	1
Retention	80	78
New business	\$ 93	\$ 65

Net written premiums for International increased \$57 million for the three months ended March 31, 2018 as compared with the same period in 2017 due to broad based growth across all of our platforms driven by higher new business, positive renewal premium change and higher retention. Excluding the effect of foreign currency exchange rates, net written premiums increased \$36 million or 14% for the three months ended March 31, 2018 as compared with the same period in 2017. The increase in net earned premiums was consistent with the trend in net written premiums.

Core income increased \$3 million for the three months ended March 31, 2018 as compared with 2017. Excluding the effect of the Federal corporate income tax rate change, core income was consistent with the prior period.

The combined ratio increased 1.5 points for the three months ended March 31, 2018 as compared with 2017. The loss ratio increased 2.1 points, primarily due to lower favorable net prior year loss reserve development. Net catastrophe losses were \$2 million, or 0.7 points of the loss ratio, for the three months ended March 31, 2018, as compared to \$3 million, or 1.7 points of the loss ratio, for the three months ended March 31, 2017. The expense ratio decreased 0.6 points for the three months ended March 31, 2018 as compared with the same period in 2017 due to higher net earned premiums.

Favorable net prior year loss reserve development was nil and \$2 million for the three months ended March 31, 2018 and 2017. Further information on net prior year loss reserve development is in Note E to the Condensed Consolidated Financial Statements included under Part I, Item 1.

The following table summarizes the gross and net carried reserves for International.

Total gross carried claim and claim adjustment expense reserves\$ 1,718\$ 1,636Net case reserves\$ 696\$ 640Net IBNR reserves782792	(In millions)	Ma	arch 31, 2018	De	cember 31, 2017
Total gross carried claim and claim adjustment expense reserves\$ 1,718\$ 1,636Net case reserves\$ 696\$ 640Net IBNR reserves782792	Gross case reserves	\$	841	\$	744
Net case reserves \$ 696 \$ 640 Net IBNR reserves 782 792	Gross IBNR reserves		877		892
Net IBNR reserves 782 792	Total gross carried claim and claim adjustment expense reserves	\$	1,718	\$	1,636
	Net case reserves	\$	696	\$	640
	Net IBNR reserves		782		792
Total net carried claim and claim adjustment expense reserves S 1,478 S 1,432	Total net carried claim and claim adjustment expense reserves	\$	1,478	\$	1,432

Life & Group

The following table details the results of operations for Life & Group.

Three months ended March 31

(In millions)	20	018	2017
Net earned premiums	\$	134	\$ 133
Net investment income		200	197
Core loss before income tax		_	(33)
Income tax benefit on core loss		14	37
Core income		14	4

Core income increased \$10 million for the three months ended March 31, 2018 as compared with the same period in 2017. Excluding the effect of the Federal corporate income tax rate change, core income increased approximately \$22 million driven by favorable persistency, and further helped by premium rate increases. The favorable persistency was driven by a high proportion of policyholders choosing to reduce benefits in lieu of premium rate increases.

Corporate & Other

The following table details the results of operations for Corporate & Other.

Three months ended March 31

(In millions)	20	18	2017
Net investment income	\$	5 \$	5
Interest expense		34	38
Core loss		(60)	(37)

Core loss increased \$23 million for the three months ended March 31, 2018 as compared with the same period in 2017. Excluding the effect of the Federal corporate income tax rate change, core loss increased approximately \$12 million due to higher adverse net prior year reserve development recorded in 2018 for A&EP under the LPT, as compared to the prior year period. This is further discussed in Note E to the Condensed Consolidated Financial Statements included under Part I, Item 1.

The following table summarizes the gross and net carried reserves for Corporate & Other.

(In millions)	 March 31, 2018	Decei	mber 31, 2017
Gross case reserves	\$ 1,210	\$	1,371
Gross IBNR reserves	 1,224		1,065
Total gross carried claim and claim adjustment expense reserves	\$ 2,434	\$	2,436
Net case reserves	\$ 93	\$	94
Net IBNR reserves	110		111
Total net carried claim and claim adjustment expense reserves	\$ 203	\$	205

INVESTMENTS

Net Investment Income

The significant components of Net investment income are presented in the following table. Fixed income securities, as presented, include both fixed maturity and non-redeemable preferred stock.

Three months ended March 31

(In millions)	2018		2017
Fixed income securities:			
Taxable fixed income securities	\$	350	\$ 348
Tax-exempt fixed income securities		105	108
Total fixed income securities		455	456
Limited partnership and common stock investments		31	90
Other, net of investment expense		4	(1)
Pretax net investment income	\$	490	\$ 545
Fixed income securities, after tax	\$	377	\$ 331
Net investment income, after tax		405	389
Effective income yield for the fixed income securities portfolio, pretax		4.7%	4.8%
Effective income yield for the fixed income securities portfolio, after tax		3.9%	3.4%

Pretax net investment income decreased \$55 million for the three months ended March 31, 2018 as compared with the same period in 2017. The decrease was driven by limited partnership investments, which returned 1.2% in 2018 as compared with 3.8% in the prior year period. However, despite the decline in limited partnership income, net investment income, after tax, increased \$16 million for the three months ended March 31, 2018 as compared with the same period in 2017 given stable fixed income returns and the lower Federal corporate income tax rate.

Net Realized Investment Gains (Losses)

The components of Net realized investment results are presented in the following table.

Three months ended March 31

(In millions)	2018	2017
Fixed maturity securities:		
Corporate bonds and other	\$ 19	\$ 30
States, municipalities and political subdivisions	20	6
Asset-backed	(21)	(4)
Total fixed maturity securities	18	32
Non-redeemable preferred stock	(15)	_
Short term and other	9	4
Net realized investment gains	12	36
Income tax expense on net realized investment gains	(2)	(11)
Net realized investment gains, after tax	\$ 10	\$ 25

Pretax net realized investment gains decreased \$24 million for the three months ended March 31, 2018 as compared with the same period in 2017. The decrease was driven by the decline in fair value of non-redeemable preferred stock and lower net realized gains on sales of securities.

Further information on our realized gains and losses, including our OTTI losses, is set forth in Note C to the Condensed Consolidated Financial Statements included under Part I, Item 1.

Portfolio Quality

The following table presents the estimated fair value and net unrealized gains (losses) of our fixed maturity securities by rating distribution.

		March 31, 2018				December 31, 2017			
(In millions)	Esti	mated Fair Value		ealized Gains Losses)	E	Estimated Fair Value	Net	Unrealized Gains (Losses)	
U.S. Government, Government agencies and Government-sponsored enterprises	\$	4,491	\$	(52)	\$	4,514	\$	21	
AAA		3,265		259		1,954		152	
AA		7,096		576		8,982		914	
A		9,168		702		9,643		952	
BBB		13,413		699		13,554		1,093	
Non-investment grade		2,826		84		2,840		140	
Total	\$	40,259	\$	2,268	\$	41,487	\$	3,272	

As of March 31, 2018 and December 31, 2017, only 2% of our fixed maturity portfolio was rated internally.

The following table presents available-for-sale fixed maturity securities in a gross unrealized loss position by ratings distribution.

	March 31, 2018		
(In millions)	Estimated Fair Value	Gross Unrealized Losses	
U.S. Government, Government agencies and Government-sponsored enterprises	\$ 2,997	\$ 82	
AAA	451	12	
AA	754	11	
A	1,579	33	
BBB	3,379	79	
Non-investment grade	881	29	
Total	\$ 10,041	\$ 246	

The following table presents the maturity profile for these available-for-sale fixed maturity securities. Securities not due to mature on a single date are allocated based on weighted average life.

	Ma	rch 31, 2018
(In millions)	Estimated Fair Value	Gross Unrealized Losses
Due in one year or less	\$ 11	\$ 4
Due after one year through five years	1,77	41
Due after five years through ten years	6,56	164
Due after ten years	1,58	37
Total	\$ 10,04	\$ 246

Duration

A primary objective in the management of the investment portfolio is to optimize return relative to corresponding liabilities and respective liquidity needs. Our views on the current interest rate environment, tax regulations, asset class valuations, specific security issuer and broader industry segment conditions and domestic and global economic conditions, are some of the factors that enter into an investment decision. We also continually monitor exposure to issuers of securities held and broader industry sector exposures and may from time to time adjust such exposures based on our views of a specific issuer or industry sector.

A further consideration in the management of the investment portfolio is the characteristics of the corresponding liabilities and the ability to align the duration of the portfolio to those liabilities and to meet future liquidity needs, minimize interest rate risk and maintain a level of income sufficient to support the underlying insurance liabilities. For portfolios where future liability cash flows are determinable and typically long term in nature, we segregate investments for asset/liability management purposes. The segregated investments support the long term care and structured settlement liabilities in the Life & Group segment.

The effective durations of fixed maturity securities, non-redeemable preferred stock and short term investments are presented in the following table. Amounts presented are net of payable and receivable amounts for securities purchased and sold, but not yet settled.

		March 31, 2018			December 31, 2017		
(In millions)	Effective Duration Estimated Fair Value (In years)		Esti	imated Fair Value	Effective Duration (In years)		
Investments supporting Life & Group	\$	16,413	8.3	\$	16,797	8.4	
Other investments		25,656	4.5		26,817	4.4	
Total	\$	42,069	6.0	\$	43,614	5.9	

The duration of the total portfolio is aligned with the cash flow characteristics of the underlying liabilities.

The investment portfolio is periodically analyzed for changes in duration and related price risk. Additionally, we periodically review the sensitivity of the portfolio to the level of foreign exchange rates and other factors that contribute to market price changes. A summary of these risks and specific analysis on changes is included in the Quantitative and Qualitative Disclosures About Market Risk included under Item 7A of our Annual Report on Form 10-K for the year ended December 31, 2017.

Short Term Investments

The carrying value of the components of the Short term investments are presented in the following table.

(In millions)	Marc	March 31, 2018		er 31, 2017
Short term investments:				
Commercial paper	\$	946	\$	905
U.S. Treasury securities		134		355
Money market funds		23		44
Other		127		132
Total short term investments	\$	1,230	\$	1,436

LIOUIDITY AND CAPITAL RESOURCES

Cash Flows

Our primary operating cash flow sources are premiums and investment income from our insurance subsidiaries. Our primary operating cash flow uses are payments for claims, policy benefits and operating expenses, including interest expense on corporate debt. Additionally, cash may be paid or received for income taxes.

For the three months ended March 31, 2018, net cash provided by operating activities was \$218 million as compared with \$282 million for the same period in 2017. The decrease in cash provided by operating activities was driven by a lower level of distributions on limited partnerships partially offset by an increase in premiums collected.

Cash flows from investing activities include the purchase and disposition of available-for-sale financial instruments and may include the purchase and sale of businesses, land, buildings, equipment and other assets not generally held for resale.

Net cash provided by investing activities was \$489 million for the three months ended March 31, 2018, as compared with \$354 million for the same period in 2017. The cash flow from investing activities is affected by various factors such as the anticipated payment of claims, financing activity, asset/liability management and individual security buy and sell decisions made in the normal course of portfolio management.

Cash flows from financing activities may include proceeds from the issuance of debt and equity securities, outflows for stockholder dividends or repayment of debt and outlays to reacquire equity securities.

For the three months ended March 31,2018, net cash used by financing activities was \$781 million as compared with \$609 million for the same period in 2017. In the first quarter of 2018, we redeemed the \$150 million outstanding aggregate principal balance of our 6.950% senior notes due January 15,2018.

Common Stock Dividends

A quarterly dividend of \$0.30 per share and a special dividend of \$2.00 per share of our common stock were declared and paid in the first quarter of 2018. On April 27, 2018, our Board of Directors declared a quarterly dividend of \$0.30 per share, payable May 30, 2018 to stockholders of record on May 14, 2018. The declaration and payment of future dividends to holders of our common stock will be at the discretion of our Board of Directors and will depend on many factors, including our earnings, financial condition, business needs and regulatory constraints.

Liquidity

Later this year, management intends to pay down the \$30 million of subordinated variable rate debt of Hardy due September 15, 2036. We believe that our present cash flows from operating, investing and financing activities are sufficient to fund our current and expected working capital and debt obligation needs and we do not expect this to change in the near term. There are currently no amounts outstanding under our \$250 million senior unsecured revolving credit facility and no borrowings outstanding through our membership in the FHLBC.

Dividends from CCC are subject to the insurance holding company laws of the State of Illinois, the domiciliary state of CCC. Under these laws, ordinary dividends, or dividends that do not require prior approval by the Illinois Department of Insurance (the Department), are determined based on the greater of the prior year's statutory net income or 10% of statutory surplus as of the end of the prior year, as well as timing and amount of dividends paid in the preceding twelve months. Additionally, ordinary dividends may only be paid from earned surplus, which is calculated by removing unrealized gains from unassigned surplus. As of March 31, 2018 CCC was in a positive earned surplus position. The maximum allowable dividend CCC could pay during 2018 that would not be subject to the Department's prior approval is \$1,073 million, less dividends paid during the preceding twelve months measured at that point in time. CCC paid dividends of \$280 million during the nine months ended December 31, 2017 and \$670 million during the three months ended March 31, 2018 . As of March 31, 2018 CCC is able to pay approximately \$123 million of dividends that would not be subject to prior approval of the Department. The actual level of dividends paid in any year is determined after an assessment of available dividend capacity, holding company liquidity and cash needs as well as the impact the dividends will have on the statutory surplus of the applicable insurance company.

We have an effective automatic shelf registration statement under which we may publicly issue debt, equity or hybrid securities from time to time.

ACCOUNTING STANDARDS UPDATE

For a discussion of Accounting Standards Updates adopted in the current period and that will be adopted in the future, see Note A to the Condensed Consolidated Financial Statements included under Part I. Item 1.

FORWARD-LOOKING STATEMENTS

This report contains a number of forward-looking statements which relate to anticipated future events rather than actual present conditions or historical events. These statements are made pursuant to the safe harbor provisions of the Private Securities Litigation Reform Act of 1995 and generally include words such as "believes," "expects," "intends," "anticipates," "estimates" and similar expressions. Forward-looking statements in this report include any and all statements regarding expected developments in our insurance business, including losses and loss reserves for A&EP and other mass tort claims which are more uncertain, and therefore more difficult to estimate than loss reserves respecting traditional property and casualty exposures; the impact of routine ongoing insurance reserve reviews we are conducting; our expectations concerning our revenues, earnings, expenses and investment activities; volatility in investment returns; expected cost savings and other results from our expense reduction activities; and our proposed actions in response to trends in our business. Forward-looking statements, by their nature, are subject to a variety of inherent risks and uncertainties that could cause actual results to differ materially from the results projected in the forward-looking statement. We cannot control many of these risks and uncertainties. These risks and uncertainties include, but are not limited to, the following:

Company-Specific Factors

- the risks and uncertainties associated with our insurance reserves, as outlined in the Critical Accounting Estimates and the Reserves Estimates and Uncertainties sections of our Annual Report on Form 10-K, including the sufficiency of the reserves and the possibility for future increases, which would be reflected in the results of operations in the period that the need for such adjustment is determined;
- the risk that the other parties to the transaction in which, subject to certain limitations, we ceded our legacy A&EP liabilities will not fully perform their obligations to CNA, the uncertainty in estimating loss reserves for A&EP liabilities and the possible continued exposure of CNA to liabilities for A&EP claims that are not covered under the terms of the transaction;
- the performance of reinsurance companies under reinsurance contracts with us; and
- the risks and uncertainties associated with potential acquisitions and divestitures, including the consummation of such transactions, the successful integration of acquired operations and the potential for subsequent impairment of goodwill or intangible assets.

Industry and General Market Factors

- the impact of competitive products, policies and pricing and the competitive environment in which we operate, including changes in our book of business;
- product and policy availability and demand and market responses, including the level of ability to obtain rate increases and decline or non-renew underpriced accounts, to achieve premium targets and profitability and to realize growth and retention estimates;
- general economic and business conditions, including recessionary conditions that may decrease the size and number of our insurance customers and create
 additional losses to our lines of business, especially those that provide management and professional liability insurance, as well as surety bonds, to businesses
 engaged in real estate, financial services and professional services and inflationary pressures on medical care costs, construction costs and other economic
 sectors that increase the severity of claims;
- conditions in the capital and credit markets, including continuing uncertainty and instability in these markets, as well as the overall economy, and their impact on the returns, types, liquidity and valuation of our investments;
- · conditions in the capital and credit markets that may limit our ability to raise significant amounts of capital on favorable terms; and
- the possibility of changes in our ratings by ratings agencies, including the inability to access certain markets or distribution channels and the required collateralization of future payment obligations as a result of such changes, and changes in rating agency policies and practices.

Regulatory Factors

- regulatory initiatives and compliance with governmental regulations, judicial interpretations within the regulatory framework, including interpretation of
 policy provisions, decisions regarding coverage and theories of liability, legislative actions that increase claimant activity, trends in litigation and the outcome
 of any litigation involving us and rulings and changes in tax laws and regulations;
- regulatory limitations, impositions and restrictions upon us, including with respect to our ability to increase premium rates, and the effects of assessments and
 other surcharges for guaranty funds and second-injury funds, other mandatory pooling arrangements and future assessments levied on insurance companies;
 and
- regulatory limitations and restrictions, including limitations upon our ability to receive dividends from our insurance subsidiaries, imposed by regulatory authorities, including regulatory capital adequacy standards.

Impact of Catastrophic Events and Related Developments

- weather and other natural physical events, including the severity and frequency of storms, hail, snowfall and other winter conditions, natural disasters such as hurricanes and earthquakes, as well as climate change, including effects on global weather patterns, greenhouse gases, sea, land and air temperatures, sea levels, rain, hail and snow;
- regulatory requirements imposed by coastal state regulators in the wake of hurricanes or other natural disasters, including limitations on the ability to exit
 markets or to non-renew, cancel or change terms and conditions in policies, as well as mandatory assessments to fund any shortfalls arising from the inability
 of quasi-governmental insurers to pay claims;
- man-made disasters, including the possible occurrence of terrorist attacks, the unpredictability of the nature, targets, severity or frequency of such events, and the effect of the absence or insufficiency of applicable terrorism legislation on coverages; and
- the occurrence of epidemics.

Referendum on the United Kingdom's Membership in the European Union

in 2016, the United Kingdom (U.K.) held a referendum in which voters approved an exit from the European Union (E.U.), commonly referred to as "Brexit." As a result of the referendum, in 2017 the British government formally commenced the process to leave the E.U. and began negotiating the terms of treaties that will govern the U.K.'s future relationship with the E.U. Although the terms of any future treaties are unknown, we believe changes in our international operating platform will be required to allow us to continue to write business in the E.U. after the completion of Brexit. Therefore, we have begun the process of establishing a new European subsidiary in Luxembourg. As a result of these changes, the complexity and cost of regulatory compliance of our European business is likely to increase.

Our forward-looking statements speak only as of the date of the filing of this Quarterly Report on Form 10-Q and we do not undertake any obligation to update or revise any forward-looking statement to reflect events or circumstances after the date of the statement, even if our expectations or any related events or circumstances change.

Item 3. Quantitative and Qualitative Disclosures About Market Risk

There were no material changes in our market risk components for the three months ended March 31, 2018. See the Quantitative and Qualitative Disclosures About Market Risk included in Item 7A of our Annual Report on Form 10-K for the year ended December 31, 2017 for further information. Additional information related to portfolio duration is discussed in the Investments section of our Management's Discussion and Analysis of Financial Condition and Results of Operations included in Part I, Item 2.

Item 4. Controls and Procedures

The Company maintains a system of disclosure controls and procedures which are designed to ensure that information required to be disclosed by the Company in reports that it files or submits to the Securities and Exchange Commission under the Securities Exchange Act of 1934, as amended (the "Exchange Act"), including this report, is recorded, processed, summarized and reported on a timely basis. These disclosure controls and procedures include controls and procedures designed to ensure that information required to be disclosed under the Exchange Act is accumulated and communicated to the Company's management on a timely basis to allow decisions regarding required disclosure.

As of March 31, 2018, the Company's management, including the Company's Chief Executive Officer (CEO) and Chief Financial Officer (CFO), conducted an evaluation of the effectiveness of the Company's disclosure controls and procedures (as such term is defined in Exchange Act Rules 13a-15(e) and 15d-15(e)). Based on this evaluation, the CEO and CFO have concluded that the Company's disclosure controls and procedures are effective as of March 31, 2018.

There has been no change in the Company's internal control over financial reporting (as defined in Rules 13a-15 (f) and 15d-15(f) under the Exchange Act) during the quarter ended March 31, 2018 that has materially affected, or is reasonably likely to materially affect, the Company's internal control over financial reporting.

PART II. Other Information

Item 1. Legal Proceedings

Information on our legal proceedings is set forth in Note F to the Condensed Consolidated Financial Statements included under Part I, Item 1.

Item 6. Exhibits

See Exhibit Index.

SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

		CNA Financial Corporation	
Dated: April 30, 2018	Ву	/s/ D. Craig Mense	
		D. Craig Mense Executive Vice President and Chief Financial Officer	
	58		

<u>Table of Contents</u>

EXHIBIT INDEX

Description of Exhibit	Exhibit Number
2018 CNA Non-qualified Savings Plan, Restated as of January 1, 2018	10.1
Certification of Chief Executive Officer	31.1
Certification of Chief Financial Officer	31.2
Written Statement of the Chief Executive Officer of CNA Financial Corporation Pursuant to 18 U.S.C. Section 1350 (As adopted by Section 906 of the Sarbanes-Oxley Act of 2002)	32.1
Written Statement of the Chief Financial Officer of CNA Financial Corporation Pursuant to 18 U.S.C. Section 1350 (As adopted by Section 906 of the Sarbanes-Oxley Act of 2002)	32.2
XBRL Instance Document	101.INS
XBRL Taxonomy Extension Schema	101.SCH
XBRL Taxonomy Extension Calculation Linkbase	101.CAL
XBRL Taxonomy Extension Definition Linkbase	101.DEF
XBRL Taxonomy Label Linkbase	101.LAB
YRRI Tayonomy Extension Presentation Linkhase	101 PRE

FXI	Π	D	\mathbf{T}	10	1



Restated as of January 1, 2018

CNA NON-QUALIFIED SAVINGS PLAN

TABLE OF CONTENTS

ARTIC	LE I <u>GENERAL PROVISIONS</u>	1
1.1	<u>Purpose</u>	1
1.2	Effective Date	1
1.3	Company and Employers	1
1.4	<u>Plan Year</u>	1
1.5	<u>Definitions and Rules of Construction</u>	1
ARTICI	LE II <u>ELIGIBILITY AND BENEFITS</u>	5
2.1	Eligibility	5
2.2	Elective Deferrals	6
2.3	Employer Contributions	7
2.4	<u>Earnings</u>	8
2.5	<u>Vesting</u>	9
2.6	Time and Form of Payment	9
2.7	Death Benefits	11
2.8	Excess Benefit Plan Participants	11
2.9	Former Participants in Surety Plans	12
ARTICI	LE III <u>PAYMENT OF BENEFITS</u>	14
3.1	Source of Payment	14
3.2	Establishment of Trust	14
3.3	Withdrawals for Financial Emergency	14
3.4	Withholding and Payroll Taxes	15
3.5	Payment on Behalf of Disabled or Incompetent Persons	15
3.6	Missing Participants or Beneficiaries	16
3.7	Other Permitted Distributions	16
3.8	Recovery of Erroneous Distributions	16
ARTICI	LE IV <u>ADMINISTRATION</u>	17
4.1	Administrator	17
4.2	Administrator's Powers	17
4.3	Binding Effect of Rulings	18
	Claims Procedure	18
4.5	<u>Indemnity</u>	20

ARTICLE V AMENDMENT AND TERMINATION OF PLAN	21
5.1 <u>Amendment</u>	21
5.2 <u>Termination</u>	21
ARTICLE VI MISCELLANEOUS	22
6.1 Status of Plan	22
6.2 Nonassignability	22
6.3 No Contract of Employment	22
6.4 Participant Litigation	22
6.5 Participant and Beneficiary Duties	23
6.6 Governing Law	23
6.7 <u>Validity</u>	23
6.8 Notices	23
6.9 <u>Successors</u>	23
APPENDIX A FULL VESTING OF PARTICIPANT AFFECTED BY CERTAIN EVENTS	25
APPENDIX B EXCESS BENEFIT PLAN	26

CNA NON-QUALIFIED SAVINGS PLAN ARTICLE I GENERAL PROVISIONS

- 1.1 <u>Purpose</u>. The purpose of this CNA Non-Qualified Savings Plan (the "Plan") is to enable selected Employees and former Employees of CNA Financial Corporation (the "Company") or its subsidiaries (the "Employers") to elect to defer additional compensation, and receive additional matching and other Company contributions, to compensate them for the limitations imposed upon their benefits under the CNA 401(k) Plus Plan in order to comply with the requirements of the Internal Revenue Code (the "Code"), and also to permit the Employers to provide additional amounts of deferred compensation for other key Employees and former Employees. The Plan was originally adopted jointly by the Company and Continental Casualty Company, one of the Employers, effective as of January 1, 1987, under the name CNA Employees' Supplemental Savings Plan, and has been amended from time to time. The Plan was most recently restated as of January 1, 2009, pursuant to which restatement the Company was granted the authority to adopt further amendments to the Plan. The Plan is hereby further amended to incorporate certain amendments made since the last restatement, and to make other changes.
- 1.2 <u>Effective Date</u>. The Plan was originally effective as of January 1, 1987, and was most recently restated as of January 1, 2009, in order to comply with the final regulations under §409A of the Code. This amendment and restatement of the Plan shall be effective as of January 1, 2018. Except as otherwise explicitly provided below, the rights of a Participant whose employment terminated, or who otherwise became entitled to receive benefits, under the Plan prior to January 1, 2018, shall be determined under the terms of the Plan as in effect at such time.
- 1.3 <u>Company and Employers</u>. The Plan is adopted for the benefit of selected Employees and former Employees of the Company and the Employers. As of the effective date of this restatement, Continental Casualty Company is the only Employer participating in the Plan. The Administrator may permit any other company that is an affiliate or subsidiary of the Company to participate in the Plan in such manner as the Administrator may determine. Each Employer is liable for the payment of benefits to a Participant that is or was an Employee of such Employer. The Company is the sponsor of the Plan for purposes of ERISA and the issuer of all interests in the Plan for securities laws purposes.
- 1.4 <u>Plan Year</u>. The Plan Year of the Plan shall coincide with the calendar year, except as the Administrator shall otherwise determine.
- 1.5 <u>Definitions and Rules of Construction</u>. As used in this Plan, certain capitalized terms shall have the meanings set forth below. Capitalized terms not defined herein shall have the meaning set forth in the CNA 401(k) Plus Plan, if applicable. Nouns and pronouns which are of one gender shall be construed to include all genders, and the singular shall include the plural and vice-versa, except as the context otherwise clearly requires. Article and Section headings are for ease of reference only and shall have no substantive meaning.

- (a) "Account" means the separate bookkeeping account maintained on the books of a Participant's Employer to reflect the amount owed to him pursuant to this Plan. Each Account shall be divided into the following subaccounts:
 - (i) The Deferred Account shall include the amounts deferred by the Participant pursuant to Section 2.2 and the income attributable thereto.
 - (ii) The Matching Account shall include any amounts credited to the Participant pursuant to Section 2.3(a) or (b) and the income attributable thereto.
 - (iii) The Employer Account shall include any amounts credited to the Participant pursuant to Section 2.3(c) and the income attributable thereto.

Each Account of each Participant who participated in the Plan prior to January 1, 2005, shall be divided into a Pre-2005 and a Post-2004 portion, as follows:

- (iv) The Pre-2005 portion of the Deferred Account shall consist of all amounts allocated to the Deferred Account on or before December 31, 2004, and any earnings thereon.
- (v) The Pre-2005 portion of the Matching Account and the Employer Account shall consist of the vested portions of such Accounts as of December 31, 2004, and any earnings thereon.
- (vi) The Post-2004 portion of each Account shall consist of any amount not included in the Pre-2005 Portion.

The Administrator may establish additional subaccounts within a Participant's Account, or may combine two or more subaccounts. The term "Account", when not otherwise specified, shall refer collectively to all of the subaccounts comprising a Participant's Account, and the terms "Pre-2005 Account" and "Post-2004 Account" shall mean, respectively, the Pre-2005 and Post-2004 Portions of a Participant's Accounts.

If a Participant participates in the Plan both as an Employee and subsequently as a former Employee, he or she shall have two separate Accounts, and any election made by him with respect to one Account shall have no effect on the other Account.

- (b) "Administrator" means the Company or such other person as the Company shall designate pursuant to Section 4.1.
- (c) "Beneficiary" means the person or persons designated to receive the Participant's Account in the event of his or her death pursuant to Section 2.7.
 - (d) "Board" means the Board of Directors of the Company.

- (e) "Code" means the Internal Revenue Code of 1986, and any treasury regulations, rulings or other authoritative administrative pronouncements interpreting the Code. If any provision of the Code specifically referred to herein is amended or replaced, the reference shall be deemed to be to the provision as so amended, or to the new provision, if such reference is consistent with the purposes of the Plan.
- (f) "Company" means CNA Financial Corporation, and any successor thereto that assumes the obligations of the Company under this Plan.
- (g) "Compensation" means Compensation as defined in Section 2.1(j) of the CNA 401(k) Plus Plan for purposes of determining a Participant's Before-Tax, After-Tax and Matching Contributions, but without regard to any limits on includable compensation imposed by the Tax Limits.
- (h) "Controlled Group" means the Company and all other entities that are part of a controlled group of corporations, or group of trades or businesses under common control, that includes the Company as defined in §414(b) or (c) of the Code; including, for avoidance of doubt, Loews Corporation and its respective 80% owned subsidiaries.
- (i) "Deferral Agreement" means an agreement between an Active Participant and his or her Employer specifying that a portion of his or her Compensation shall be withheld and credited to his or her Account in the Plan pursuant to Section 2.2, or providing that additional amounts will be credited to his or her Account pursuant to Section 2.3, or both, and any amendment thereto. To the extent determined by the Administrator, a Deferral Agreement may take the form of an election made by the Participant either in writing or through electronic communications, and a Participant's election to participate in the CNA 401(k) Plus Plan may be treated as a Deferral Agreement under this Plan in the absence of a contrary election. The term "Deferral Agreement" may also refer to any provision of an employment, consulting, severance, or other agreement for the performance of services that makes specific reference to this Plan and provides for deferred compensation.
- (j) "Employee" means any person employed by any Employer and classified as an Employee by such Employer. Except as otherwise provided in Section 2.1(c), the term "Employee" shall not include a person who is retained to provide services for an Employer as an independent contractor, or who provides services for an Employer pursuant to an agreement or understanding, written or unwritten, with a third party that such person shall be treated as an employee of the third party, but who is subsequently determined to be an employee at common law, for purposes of any federal or state tax or employment law, or for any other purpose.
- (k) "Employer" means any subsidiary of the Company that adopts the Plan and is the employer or former employer of a Participant.
- (l) "ERISA" means the Employee Retirement Income Security Act of 1974, and any Labor Department regulations, rulings or other authoritative administrative pronouncements interpreting ERISA. If any provision of ERISA specifically referred to herein is amended or

replaced, the reference shall be deemed to be to the provision as so amended, or to the new provision, if such reference is consistent with the purposes of the Plan.

- (m) Excess Benefit Plan" means the separable portion of the Plan contained in Appendix B.
- (n) "Participant" means an Employee or former Employee designated to participate in the Plan pursuant to Section 2.1, while he or she has the right to any benefits under the Plan. Participants are divided in Active Participants and Inactive Participants, as described in Section 2.1, and the term "Participant", when not modified, shall refer to both Active and Inactive Participants, unless clearly inconsistent with the context.
 - (o) "Plan" means this CNA Non-Qualified Savings Plan, as amended from time to time.
- (p) "Tax Limits" means the limitations imposed on a Participant's benefits under the Plan to satisfy the requirements of §401(a)(17), §402(g), or §415 of the Code.
- (q) "Total Pay" means Total Pay as defined in the 401(K) Plus Plan (and as formerly referred to as "Retirement Plan Compensation") for purposes of determining a Participant's Basic and Performance Contributions, but without regard to any limits on includible compensation imposed by the Tax Limits.
- (r) "401(K) Plus Plan" means the CNA 401(k) Plus Plan, as amended from time to time, and, if appropriate, any new plan adopted by the Company to replace the 401(K) Plus Plan. In the case of a Participant who participates in a plan maintained by his or her Employer other than the CNA 401(k) Plus Plan, which plan is qualified under §401(a) of the Code and includes a cash or deferred feature qualified under §401(k) of the Code, the term "401(K) Plus Plan" with respect to such Participant shall mean such other plan.

ARTICLE II ELIGIBILITY AND BENEFITS

2.1 Eligibility

- (a) Only selected management and highly compensated Employees and former Employees who are designated as provided herein shall be eligible to participate in the Plan. The Employees and former Employees who are so designated to participate in the Plan shall be referred to herein as "Active Participants" for so long as they have the right to have additional amounts credited to their Accounts pursuant to Section 2.2 or 2.3. A person who is no longer an Active Participant, but who still has an undistributed Account in the Plan, shall be referred to as an "Inactive Participant."
- (b) Unless otherwise determined by the Administrator, only the following Employees who are eligible to participate in the 401(K) Plus Plan are eligible to participate in the Plan:
 - (i) An Employee whose Compensation for the Plan Year exceeds (or, as determined by the Administrator, is expected to exceed) the limitation of Code §401(a)(17);
 - (ii) An Employee hired during the Plan Year with a base salary that exceeds the limitation of Code §401(a)(17) (without regard to whether the Employee's total Compensation for the Plan Year is expected to exceed such limitation) shall be eligible to participate on his or her date of hire; provided that such Employee has not participated (other than through accrual of earnings on amounts previously deferred) in any account balance nonqualified deferred compensation arrangement sponsored by the Company or any member of the Controlled Group during the 24 month period prior to the date he or she is hired, unless the employee received a distribution of his or her entire balance in such plan during such 24 month period, and immediately prior to such distribution was not eligible to continue to participate in such plan. An Employee whose Compensation unexpectedly exceeds the limitation of §401(a)(17) during a Plan Year, and who otherwise satisfies the requirements of the preceding sentence (including the receipt of any Basic or Performance Contributions under this Plan during such 24 month period) may, if permitted by the Administrator, be treated as have been hired on the date that his or her Compensation exceeds such limit; and
 - (iii) An Employee shall be eligible to participate, solely for purposes of being credited with Basic and Performance Contributions, in the first Plan Year in which his or her Total Pay exceeds the limitation of Code §401(a)(17).

Notwithstanding the foregoing, the Administrator may, in its sole discretion, determine at any time that any Employee or group of Employees described in this paragraph (b) shall no longer be eligible to participate.

- (c) Any Employer, with the consent of the Administrator, may enter into a Deferral Agreement with a person not described in paragraphs (a) or (b), who may be either an Employee or a former Employee, and such person shall thereby become an Active Participant. To the extent necessary or appropriate, any reference in this Plan to "employment" shall be modified and interpreted in the case of a former Employee or independent consultant in a manner consistent with the intent of the Plan.
- (d) Any Employee who becomes a Participant, but who becomes ineligible to enter into a Deferral Agreement for any subsequent Plan Year by reason of a decrease in Compensation, shall remain a Participant, and shall be credited with Employer Contributions pursuant to Section 2.3, for so long as he remains an Employee, unless otherwise determined by the Administrator.

2.2 <u>Elective Deferrals</u>.

- (a) Each Active Participant may, for any Plan Year in which he or she is also a participant in the 401(K) Plus Plan, elect in his or her Deferral Agreement to accept a reduction in his or her Compensation from his or her Employer equal to a whole percentage not to exceed the 50% of his or her Compensation. At present, the terms of the 401(K) Plus Plan do not permit a Participant who is participating in the 401(K) Plus Plan during a Plan Year to change his or her Before-Tax or Roth 401(k) Deferral elections during the Plan Year; however, if the 401(K) Plus Plan is amended or terminated, or if for any other reason a Participant is permitted to change his or her 401(K) Plus Plan Before-Tax or Roth 401(k) Deferral election during a Plan Year, the percentage withheld and credited to the Participant's Deferral Account shall be calculated as if the 401(K) Plus Plan Before-Tax or Roth 401(k) Deferral election had not changed. Eligible Participants may make an excess deferral election that does not go into effect until contributions to the 401(K) Plus Plan have been discontinued because of a Tax Limit. Effective January 1, 2018, simultaneous deferral contribution elections were eliminated.
- (b) All deferral elections shall be made in accordance with procedures established by the Administrator during the periods described below, and shall be irrevocable after the end of the period during which the election may be made. Except as otherwise provided in procedures established by the Administrator, a deferral election for one Plan Year shall apply to all future Plan Years unless changed by the Participant during the applicable election period:
- (i) Except as otherwise provided below, all deferral elections shall be made not later than the last day of the Plan Year immediately preceding the Plan Year to which the deferral election shall apply.
- (ii) An Employee who first becomes eligible to participate during a Plan Year pursuant to Section 2.1(b)(ii) may make a deferral election not later than 30 days after he or she becomes eligible, which deferral election shall apply only to Compensation earned after the date of the election.

- (c) Any Employer, with the consent of the Administrator, may enter into a Deferral Agreement with an Active Participant (including but not limited to a person described in Section 2.1(b)) which provides for Compensation to be withheld and credited to the Active Participant's Deferral Account on a basis different from that described in paragraph (a). Such a Deferral Agreement may provide for the deferral of forms or amounts of compensation different from those defined as Compensation in Section 1.5(h), including payments to a former Employee or independent contractor, in which event such compensation shall be considered Compensation for all purposes of this Plan. Notwithstanding the foregoing, effective January 1, 2005, if any Deferral Agreement permits a Participant to defer any form of incentive compensation, as defined in Code §409A, that is measured over a period of twelve months or more, the deferral election must be made not less than six months before the end of the measurement period.
- (d) Amounts deferred pursuant to paragraph (a) shall be credited to the Active Participant's Deferral Account as of the date on which the deferred Compensation would otherwise have been paid. No election, and no provision of any Deferral Agreement, shall permit a Participant to defer Compensation already earned when the election is made. Effective January 1, 2005, all deferral elections, including those under a Deferral Agreement, must be made not later than December 31 of the immediately preceding year (except as otherwise provided in paragraph (b)(ii), or in paragraph (c) with respect to deferrals of incentive compensation), and may thereafter be revoked or modified only as permitted in regulations issued pursuant to Code §409A.

2.3 <u>Employer Contributions</u>.

- (a) For each payroll period, the Employer of an Active Participant shall credit to the Active Participant's Matching Account an amount equal to the amount deferred by the Active Participant for such payroll period under Section 2.2 multiplied by the Fixed Matching Contribution percentage applicable to such Active Participant under the 401(K) Plus Plan. The Company shall also credit to the Matching Account of an Active Participant any Fixed Matching Contribution that relates to a Before-Tax and/or Roth 401(k) contribution made under the 401(K) Plus Plan, but which Fixed Matching Contribution cannot be allocated to such Active Participant's 401(K) Plus Plan account without exceeding the Tax Limits. The total amount of Fixed Matching Contributions credited to an Active Participant's Matching Account under this Section 2.3 for each Plan Year shall not exceed the excess of 6% the Active Participant's total Compensation for the Plan Year reduced by all Fixed Matching Contributions allocated to his or her account in the 401(K) Plus Plan for the same Plan Year.
- (b) In addition to the amounts set forth above, at the end of each Plan Year the Employer of an Active Participant shall credit to the Active Participant's Matching Account an amount equal to the amount deferred by the Active Participant for the Plan Year pursuant to Section 2.2, multiplied by the Variable Matching Contribution percentage applicable to such Active Participant under the 401(K) Plus Plan. The Company shall also credit to the Matching Account of an Active Participant any Variable Matching Contribution that relates to a Before-Tax and/or Roth 401(k) Contribution made under the 401(K) Plus Plan, but which Matching Contribution cannot be allocated to such Active Participant's 401(K) Plus Plan account without exceeding the Tax Limits.

- (c) In addition to the amounts set forth above, at the end of each Plan Year or pay period, as applicable, the Employer of an Active Participant shall credit to the Active Participant's Employer Contribution Account an amount equal to the portion of the Active Participant's Total Pay that exceeds the Tax Limits multiplied by the applicable Basic and Performance Contribution percentages applicable to such Active Participant under the 401(K) Plus Plan. The Company shall also credit to the Employer Contribution Account of an Active Participant any Basic or Performance Contribution that cannot be allocated to such Active Participant's 401(K) Plus Plan account without exceeding the Tax Limits.
- (d) Anything else contained herein to the contrary notwithstanding, the amount credited to an Active Participant's Matching Account or Employer Contribution Account pursuant to paragraph (a), (b) or (c) for any Plan Year shall not exceed the amount of additional Fixed Matching, Variable Matching, Basic or Performance Contributions, as the case may be, that would have been allocated to the Active Participant's 401(K) Plus Plan account for the same Plan Year if the Tax Limits did not apply.
- (e) Any Employer, with the consent of the Administrator, may enter into an employment agreement, or adopt employment policies, with or applicable to an Active Participant (including but not limited to a person described in Section 2.1(b)) which provides for amounts to be credited to the Active Participant's Matching or Employer Account on a basis different from that described in paragraph (a), (b) or (c). Such an agreement or policy shall specify the basis upon which the amount to be so credited shall be determined, and may also specify a vesting schedule different than that specified in Section 2.5.

2.4 Earnings.

- (a) Except as otherwise provided in paragraph (b), earnings shall be credited to each Participant's Account at the projected rate of return on the Invesco Stable Value Fund established under the 401(K) Plus Plan. In the event that the Invesco Stable Value Fund is no longer offered as an investment alternative under the 401(K) Plus Plan, the Administrator shall designate a reasonably equivalent investment option under the 401(K) Plus Plan to be used to measure the rate at which earnings shall be credited.
- (b) At any time after the effective date of this restatement, the Administrator may designate selected mutual funds or other investment media ("funds"), and each Participant shall have the right to have earnings (including realized and unrealized gains and losses) on his or her Account computed as if it had been invested in such funds in such proportions as the Participant shall elect. The funds may be the same as the Investment Funds designated under the 401(K) Plus Plan, or may exclude some or all of such Investment Funds or include other funds as the Administrator may determine. The portion of each Participant's Account that is deemed to be invested in each fund shall be a whole percentage, and elections may be changed at such intervals and in such manner as the Administrator may determine. The Administrator shall have the authority to select and discontinue funds at any time, to establish a rate at which interest shall be credited on Accounts with respect to which no fund election is in effect, and otherwise to establish rules and

procedures with respect to the calculation and crediting of earnings, including changing the intervals at which fund elections may be made or at which earnings are posted, and establishing a minimum or maximum percentage that may be deemed invested in any fund.

- (c) Anything else contained herein to the contrary, in no event shall any Participant be allowed to elect a rate of return on his or her Account retroactively, and in all cases earnings shall be computed in such a manner that they shall not be considered additional deferred compensation for purposes of FICA withholding under §3121(v) of the Code.
- 2.5 <u>Vesting</u>. The balance in a Participant's Deferral Account shall be fully vested and nonforfeitable at all times. The balance in a Participant's Matching Account or Employer Account (or any subaccount thereof) shall be vested at the same times and to the same extent as the Participant's analogous account in the 401(K) Plus Plan (except as otherwise provided in a Deferral Agreement with respect to amounts credited pursuant to Section 2.3(b)); provided, however, that an event that results in the 401(K) Plus Plan accounts of a group of Participants being vested without regard to their years of service, including but not limited to the sale of a business unit or a determination that a partial termination of the 401(K) Plus Plan has occurred, shall apply to this Plan if and only if such event is listed in Appendix A to this Plan. To the extent a Participant's Account is not vested at the time of his or her termination of employment for any reason, the non-vested portion shall be forfeited, and neither the Company nor any Employer shall have any further obligation to him whatsoever with respect to the forfeited portion.

2.6 <u>Time and Form of Payment</u>.

- (a) Except as otherwise provided in paragraph (c), the vested balance in a Participant's Account shall be paid to the Participant in one of the following manners, as elected by the Participant in accordance with paragraph (b):
 - (i) In a single lump sum, paid as soon as practical, but in no event more than 90 days following the Participant's termination of employment;
 - (ii) In a single lump sum, paid during January of the year following the year that includes the Participant's termination of employment; or
 - (iii) In a series of not more than 10 annual installments, payable during January of each year commencing with the year following the year that includes the Participant's termination of employment. Each such installment shall be equal to the remaining balance in Participant's Account immediately prior to the payment of such installment divided by the number of installments remaining to be paid.
- (b) The time and method of payment shall be elected by the Participant in accordance with the procedures established by the Administrator at the earlier of the following times:
 - (i) When the Participant enters into his or her first Deferral Agreement or,

- (ii) In the case of a newly eligible Participant, not later than 30 days after the date of the paycheck that causes his or her Total Pay to exceed the §401(a)
 - (17) limit in the first year in which his or her Total Pay exceeds such limit.

If the Participant does not specify a time and method of payment, the vested balance in his or her Account shall be distributed in a single lump sum as soon as administratively feasible, but not more than 90 days, following his or her termination of employment. The initial election or deemed election as to the time and form of distribution cannot be changed.

- (c) Anything else in this Plan, or a Deferral Agreement, to the contrary notwithstanding:
- (i) Except as otherwise provided below, no part of a Participant's Post-2004 Account shall be payable until the Participant has incurred a separation from service as defined in Code §409A.
- (ii) No part of a Participant's Post-2004 Account shall be payable to a Participant who is a specified employee, as defined in Code §409A, until the first business day that is at least six months after he or she has incurred a separation from service. The identification of Participants as specified employees shall be made as of December 31 of each year by Loews Corporation based upon the employees of the controlled group of which Loews Corporation is the common parent, and a Participant identified as a specified employee as of any December 31 shall be subject to the provisions of this paragraph (c)(ii) if the Participant incurs a separation from service during the twelve month period commencing on the following April 1.
- (iii) In no event shall the distribution of any Post-2004 Account be accelerated to a time earlier than which it would otherwise have been paid, whether by amendment of the Plan, exercise of the Employee Benefits Committee's discretion, or otherwise, except as permitted by regulations issued pursuant to Code §409A.
- (iv) In the event that the Administrator, in its sole discretion, determines that any time or form of distribution provided for in the Plan, or the existence of a right to elect a different time or form of distribution, would cause the Plan to fail to meet the requirements of Code §409A, or otherwise cause Participants to be subject to any adverse federal income tax consequences, the Administrator shall adopt procedures modifying or removing the form of distribution or election right, which shall be deemed an amendment to the Plan.
- (v) Any Deferral Agreement that provides for a different form or time of payment shall specify the time and manner of payment, without Employer or Participant discretion, at the time the Deferral Agreement is entered into, and shall otherwise comply with the requirements of this paragraph (b); provided

that, in addition to a severance from service, a Deferral Agreement may provide for benefits to be paid at a specified time or pursuant to a fixed schedule set forth in the Deferral Agreement, or upon the occurrence of a change in ownership or control of the Participant's Employer, or in a substantial portion of its assets, as defined in Code §409A, and provided further that a Deferral Agreement may permit a Participant to elect to further defer the distribution of his or her Account if the election does not take effect for at least twelve months and the distribution of the Post-2004 Account is deferred by at least five years.

2.7 Death Benefits.

- (a) If a Participant dies while still employed, his or her Account shall be fully vested and shall be paid to his or her Beneficiary in a single lump sum. If a Participant dies after his or her employment has been terminated but before his or her Account has been paid in full, the remaining balance in his or her Account shall be paid to his or her Beneficiary in a single lump sum, regardless of whether the Participant has elected payment in installments (and regardless of whether the Participant was a specified employee at the time of his separation from service as provided in Section 2.6(c)(ii)). All payments to Beneficiaries shall be within 90 days following the Participant's death.
- (b) A Participant's Beneficiary shall be the person or persons designated by the Participant in his or her Deferral Agreement. A Participant may change his or her Beneficiary from time to time without the consent of the Beneficiary. Subject to rules, procedures, and limitations established by the Administrator, a Beneficiary may be an entity (including a trust or nonprofit organization), and the Participant may designate multiple or contingent Beneficiaries and specify the manner in which his or her Account will be divided among them. All designations of Beneficiaries, and revocations or changes in designations, shall be made in accordance with rules, procedures and limitations prescribed by the Administrator. No designation of a Beneficiary, and no revocation or change in a designation, shall be effective until actually received by the Administrator in writing, and the Administrator's determination of a Participant's Beneficiary, if made in good faith, shall be final and conclusive on all parties.
- (c) The determination of the Participant's Beneficiary shall be made at the time of his or her death. If there is no designated Beneficiary living at the time of the Participant's death, his or her Beneficiary shall be the person designated as his or her beneficiary under the 401(K) Plus Plan, or any similar retirement plan which permits the Participant to designate a beneficiary, as determined by the Administrator in its sole discretion (regardless of whether such designation is invalid solely by reason of §401(a) (11) of the Code or Section 205 of ERISA by reason of the failure of the Participant's spouse to consent) or, if no beneficiary is designated under the 401(K) Plus Plan or any such other plan, his or her estate. If the Participant has designated more than one Beneficiary and not specified the manner in which his or her Account shall be divided, it shall be divided among all living Beneficiaries at the time of his or her death, per stirpes.
- 2.8 <u>Excess Benefit Plan Participants</u>. If an Employee who has been a Participant in the Excess Benefit Plan becomes eligible pursuant to Section 2.1, such Employee shall become a Participant, and his Account under the Excess Benefit Plan shall be transferred to and become a

part of his Account under the Plan. Such Participant shall not be eligible to make a new payment election under Section 2.6(b), but any election made by the Participant under the Excess Benefit Plan shall be treated as having been made under Section 2.6(b) and shall apply to his entire Account.

2.9 Former Participants in Surety Plans.

- (a) Effective as of December 25, 2011, the employees of Western Surety Company, a subsidiary of CNA Surety Corporation, became employees of Continental Casualty Company. CNA Surety Corporation sponsored two nonqualified deferred compensation plans for the benefit of employees of Western Surety Company: the CNA Surety Corporation Deferred Corporation Plan (the "2000 Surety Plan"), which provided for the deferral of compensation earned prior to 2005, and the CNA Surety Corporation 2005 Deferred Corporation Plan (the "2005 Surety Plan", and, collectively with the 2000 Surety Plan, the "Surety Plans"), which provides for the deferral of compensation earned beginning in 2005. Effective January 3, 2012, each of the Surety Plans was merged with and into the Plan in accordance with the provisions of this Section 2.9.
- (b) Effective January 3, 2012, each person who has an account in either of the Surety Plan (a "Surety Participant") became a Participant. Each Surety Participant's account in the applicable Surety Plan, valued as of such date, became an Account in the Plan, and the Plan hereby assumes the obligation to pay such Account, subject to the terms of the Plan. If a Surety Participant had an account in the 2005 Surety Plan, such account became his Account in this Plan, any future contributions shall be credited to such Account, and the entire balance in such Account shall be considered a Post-2004 Account for all purposes of the Plan. If a Surety Participant has an account in the 2000 Surety Plan, whether or not he also has an account in the 2005 Surety Plan, such account shall be treated as a separate Account or subaccount which is a Pre-2005 Account for all purposes of the Plan, and no further contributions shall be allocated to such Account.
 - (c) Contributions to the Plan on behalf of Surety Participants were determined as follows:
 - (i) Elective Contributions, Matching Contributions, and Basic Contributions for the final Western Surety Company payroll period ending prior to December 25, 2011, were calculated in accordance with the terms of the 2005 Surety Plan, credited initially to the Surety Participant's account in the 2005 Surety Plan, and were transferred to the Plan on January 3, 2012, as described in Section 2.9(b) above.
 - (ii) Performance Contributions for 2011 were calculated in accordance with the terms of the 2005 Surety Plan (based upon the 2011 performance of CNA Surety Corporation), and credited to each eligible Surety Participants' Post-2004 Account in this Plan.
 - (iii) Surety Participants shall be considered Active Participants, and eligible to enter into a Deferral Agreement for 2012 and subsequent years, if and only if they otherwise satisfy the requirements of Section 2.1.

- (iv) Surety Participants who became employees of Continental Casualty Company shall be eligible to be credited with employer contributions pursuant to Section 2.3 if they are Active Participants in this Plan, or pursuant to the Excess Benefit Plan set forth in Appendix B if they are not Active Participants.
- (d) All Accounts transferred from the Surety Plans shall be credited with earnings in accordance with Section 2.4 beginning on January 3, 2012, and Surety Participants shall have no right to elect investment funds after such date.
- (e) The entire Post-2004 Account of a Surety Participant, including any amount transferred from the 2005 Surety Plan and all subsequent contributions, shall be distributed in a single lump sum as soon as reasonably practical after the date that is six months after the Surety Participant incurs a separation from service as defined in Code §409A, in accordance with Article VI of the 2005 Surety Plan.
- (f) The Pre-2005 Account of a Surety Plan that was transferred from the 2000 Surety Plan shall be distributed in accordance with the Surety Participant's election made under the terms of the 2000 Plan, in accordance with Article VI of the 2000 Surety Plan.
- (g) The right of a Surety Participant to withdraw a portion of his Account by reason of unforeseeable financial emergency pursuant to Section 5.6 of the 2000 Surety Plan or 6.5 of the 2005 Surety Plan shall be governed exclusively by Section 3.3 of the Plan. A Surety Participant whose balance in the 2000 Surety Plan is transferred to a Pre-2005 Account in this Plan shall retain the right to an early distribution pursuant to Section 5.5 of the 2000 Surety Plan, but only with respect to such Pre-2005 Account.
- (h) The Administrator may treat any Beneficiary of a Surety Participant designated pursuant to one of the Surety Plans as the Participant's Beneficiary under this Plan, subject to each Participant's right to designate a new Beneficiary in accordance with the Plan.
- (i) Surety Participants shall be fully vested in their Accounts, including both amounts transferred to the Plan from the Surety Plans, and all subsequent contributions, subject to the provisions of Sections 4.9 of the 2000 Surety Plan and 5.8 of the 2005 Surety Plan relating to forfeiture for misconduct.
- (j) Except as otherwise provided herein, Surety Participants shall be considered Participants, and the portion of their Accounts representing amounts transferred from the Surety Plans shall be treated in the same manner as other Accounts, for all purposes of the Plan. Any former employee of Western Surety who was not a Surety Participant, and becomes an employee of Continental Casualty Company on December 25, 2011 or thereafter, shall be treated in the same manner as any newly hired employee for purposes of the Plan, but shall be given credit for Vesting Service earned as an employee of Western Surety Company.

ARTICLE III PAYMENT OF BENEFITS

- 3.1 Source of Payment. All payment of benefits under the Plan shall be made directly from the general funds of the Participant's Employer. Each Employer shall establish separate bookkeeping accounts to reflect its liability under the Plan and may, but shall not be obligated to, invest in insurance or annuity contracts or other assets to assure a source of funds for the payment of benefits, but any such bookkeeping account, insurance or annuity contracts, or other investment shall constitute assets solely of such Employer, and Participants shall have no right, title or interest therein prior to payment of their benefits hereunder. The right of any Participant or other person to receive benefit payments under the provisions of this Plan shall be no greater than the right of any unsecured general creditor of the Participant's Employer. This Plan shall not create nor be construed to create a trust or fiduciary relationship in favor of any person whatsoever.
- Establishment of Trust. The Company may, but shall in no event be required to, establish one or more trusts and contribute, or cause Employers to contribute, amounts to such trusts to be used for the payment of benefits under this Plan. Any such trust shall be of the type commonly referred to as a "rabbi trust", and the Company or Employer shall be treated as the owner of the assets of such trust for tax purposes in accordance with §671-§678 of the Code. The assets of any such trust shall remain subject to the claims of creditors of the Company or the Employer contributing such assets, and no Participant or any other person shall have any beneficial interest in or other claim to the assets of any such trust beyond that of a general creditor as provided in Section 3.1. Any payments made to or on behalf of a Participant or Beneficiary from any such trust shall fully discharge the liability of the Company or Employer to such Participant or Beneficiary under the Plan to the extent of the amount so paid. The Administrator shall have the right to select, remove, and replace the trustee thereof at any time in its sole discretion, and shall enter into one or more agreements governing such trust containing such terms as it determines, and may modify, amend or revoke any such agreements, all in its sole discretion. Without limiting the generality of the foregoing, the Administrator is hereby authorized and directed to enter into an agreement with one or more trustees designated by it establishing such a trust to accept the transfer of assets from the trust established by an agreement dated August 1, 2000, between CNA Surety Corporation and First National Bank in Sioux Falls to fund the 2000 Surety Plan (as defined in Section 2.9), and from the trust established by an agreement dated March 28, 2005, between the same parties to fund the 2005 Surety Plan (as defined in Section 2.9), and to commingle such assets into a single trust fund to be used to pay any benefits accrued under the Plan, including but not limited to accounts transferred from the Surety Plans, and to pay expenses of such trust and the Plan.
- 3.3 <u>Withdrawals for Financial Emergency</u>. A Participant may withdraw part or all of the vested portion of his or her Account if the amount withdrawn is reasonably necessary to satisfy an unforeseeable financial emergency. Any such withdrawals shall be subject to such rules, procedures and limitations as the Administrator may, in its sole discretion, determine. For purposes of this Section 3.3, an unforeseeable financial emergency means a severe financial hardship to the Participant resulting from a sudden and unexpected illness or accident of the Participant, one of his or her dependents (as defined in §152(a) of the Code), or the person designated as the Participant's primary Beneficiary, loss of the Participant's property due to casualty, or other similar extraordinary

and unforeseeable circumstances arising as a result of events beyond the control of the Participant. A financial hardship that is foreseeable or within the Participant's control, such as the need or desire to purchase a residence or to send a child to college, shall not be considered an unforeseeable financial emergency. The determination of whether a Participant's need for funds constitutes an unforeseeable financial emergency shall be made in accordance with the requirements of §409A of the Code. The amount withdrawn may not exceed the amount necessary to satisfy the financial hardship (taking into account any tax payable on the withdrawal). determined after taking into account other sources of funds available to the Participant, including but not limited to reimbursement or compensation by insurance or otherwise, and the liquidation of other assets to the extent that such liquidation would not itself cause severe financial hardship. If a Participant has a financial hardship, the Participant's Deferral Agreement, if any, shall be revoked for the Plan Year (and no subsequent Deferral Agreement may be made for the same Plan Year), and the additional income resulting from such revocation shall be taken into account in determining the amount of distribution reasonably necessary to relieve the financial hardship. A Participant shall not be required to take any hardship withdrawal or loan to which he or she is entitled under the 401(K) Plus Plan or any other tax qualified retirement plan as a condition of receiving a distribution pursuant to this Section 3.3, but if a Participant receives a hardship withdrawal from the 401(K) Plus Plan or any other tax-qualified §401(k) plan maintained by an Employer and the terms of such plan require a suspension of the Participant's deferrals for six months following the date of the distribution, then the Participant's Deferral Agreement shall be permanently revoked with respect to any compensation paid or payable to the Participant during such six month period. Upon completion of the six month suspension period, the Participant's Deferral Agreement will be reinstated according to the election made during the most recent annual election period.

- 3.4 <u>Withholding and Payroll Taxes</u>. The Administrator shall withhold, or shall direct the person making any payment to withhold, from payments made hereunder any taxes required to be withheld from a Participant's wages for the federal or any state or local government. To the extent that benefits hereunder are subject to tax under the Federal Insurance Contributions Act or any other law prior to the time that they become payable, the Administrator may either withhold, or direct the Participant's Employer to withhold, the amount of such taxes from any other compensation or other amounts payable to the Participant, or may deduct the amount of such tax from the Participant's Account in accordance with Section 3.7(c). The Administrator's determination of the amount to be so withheld or deducted shall be final and binding on all parties.
- 3.5 Payment on Behalf of Disabled or Incompetent Persons. If a Plan benefit is payable to a minor or a person declared incompetent or to a person whom the Administrator, in its sole discretion, determines to be incapable of handling the disposition of property, the Administrator may direct payment of such Plan benefit to the guardian, legal representative or person having the care and custody of such minor or incompetent person, or to any other person, including any family member, whom the Administrator determines in its sole discretion to be best suited to receive and apply the payment for the benefit of such person. The Administrator may require proof of incompetency, minority, incapacity or guardianship as it may deem appropriate prior to distribution of the Plan benefit. Such distribution shall completely discharge the Company and the Participant's Employer from all liability with respect to such benefit.

- 3.6 <u>Missing Participants or Beneficiaries</u>. If the Administrator is unable to locate any Participant, Beneficiary or other person entitled to benefits under this Plan, the Administrator may, in its sole discretion, either cause all or a portion of such payment to be forfeited and to reduce its obligations under this Plan, or may pay all or a portion of such benefit to members of the missing person's family or such other person as it may determine in its sole discretion to be fair and equitable. Any payment made pursuant to this Section 3.6 shall fully discharge the obligation of the Company and all Employers under this Plan with respect to the amount so paid.
- 3.7 <u>Other Permitted Distributions</u>. Notwithstanding the foregoing provisions of this Article III, the Administrator in its sole discretion may provide for all or a portion of the balance in a Participant's Account to be distributed to the Participant, provided that no Participant may be allowed to elect to receive such a distribution:
- (a) If the total balance in a Participant's Account does not exceed the limit in effect under §402(g) of the Code, the Administrator may direct that the entire balance be distributed to the Participant in full satisfaction of his or her interest in the Plan, provided that the Participant's entire balance in all other account balance deferred compensation plans maintained by any member of the Controlled Group is also distributed to the Participant (and is taken into account in determining whether the total balance exceeds the limit in effect under §402(g)).
- (b) If any portion of a Participant's Account is determined to be includible in the Participant's taxable income by reason of the operation of §409A of the Code, the amount includible in income shall be distributed to the Participant as soon as practical.
- (c) The Administrator may direct that the Participant's portion of the FICA tax imposed on amounts deferred under the Plan pursuant to §3121 of the Code be charged to the Participant's Account, provided that the total amount charged to the Account shall not exceed the FICA tax plus income tax withholding on the amount applied to payment of the FICA tax.
- 3.8 <u>Recovery of Erroneous Distributions</u>. If the Administrator determines that the amount paid to any Participant or Beneficiary exceeded the amount that should have been paid pursuant to the terms of the Plan, the Participant or Beneficiary shall repay the amount of the excess to the Plan upon demand, and the Administrator may, on behalf of the Plan, pursue offset the amount of such excess against any other amount owed by an Employer to the Participant or Beneficiary to the maximum extent permitted by law, or pursue any other remedy available at law or equity for the recovery of such excess. Each Participant or Beneficiary who receives an excess distribution shall hold such distribution in trust for the benefit of the Plan.

ARTICLE IV ADMINISTRATION

- Administrator. This Plan shall be administered by Continental Casualty Company, which shall be the "administrator" for purposes of Section 3(16)(A) of the Employee Retirement Income Security Act of 1974. The Company may designate one or more persons who may be officers or Employees of any Employer, to exercise any of its authority or carry out any of its duties under the Plan, but such person shall not be considered the "administrator" unless specifically so designated in a resolution of the Board. In the absence of any other designation, the senior officer of Continental Casualty Company responsible for human resources, or persons acting under his or her supervision, shall be so designated. In addition, Continental Casualty Company has established an Employee Benefits Committee to oversee the operation of various retirement plans, and the Employee Benefits Committee shall have the authority on behalf of the Administrator to adopt rules, regulations and procedures, to hear all appeals from denied claims under Section 4.4, and to consider all other issues related to the administration of the Plan referred to it by senior officer of Continental Casualty Company responsible for human resources and his or her delegates.
- 4.2 <u>Administrator's Powers</u>. The Administrator shall have such powers as may be necessary to discharge its duties hereunder, including, but not by way of limitation, the following powers, rights and duties:
- (a) <u>Interpretation of Plan</u>. The Administrator shall have the power, right and duty to construe and interpret the Plan provisions and to determine all questions arising under the Plan including questions of Plan participation, eligibility for Plan benefits and the rights of Employees, Participants, Beneficiaries and other persons to benefits under the Plan and to determine the amount, manner and time of payment of any benefits hereunder.
- (b) <u>Plan Procedures</u>. The Administrator shall have the power, right and duty to adopt procedures, rules, regulations and forms to be followed by Employees, Participants, Beneficiaries and other persons or to be otherwise utilized in the efficient administration of the Plan and as are consistent with the Plan.
- (c) <u>Benefit Determinations</u>. The Administrator shall have the power, right and duty to make determinations as to the rights of Employees, Participants, Beneficiaries and other persons to benefits under the Plan and to afford any Participant or Beneficiary dissatisfied with such determination with rights pursuant to a claims procedure adopted by the Administrator in accordance with Section 4.4.
- (d) <u>Enforcement of the Plan</u>. The Administrator shall have the power, right and duty to enforce the Plan in accordance with the terms of the Plan and to enforce its procedures, rules or regulations.
- (e) <u>Maintenance of Plan Records</u>. The Administrator shall be responsible for preparing and maintaining records necessary to determine the rights and benefits of Employees, Participants and Beneficiaries or other persons under the Plan.

- (f) <u>Allocation of Duties</u>. The Administrator shall be empowered to allocate fiduciary responsibilities and the right to employ agents (who may also be Employees of the Company) and to delegate to them any of the administrative duties imposed upon the Administrator.
- (g) <u>Correction of Errors</u>. To correct any errors made in the computation of benefits under the Plan, and, if a trust has been established, to recover any contributions made to such trust by mistake of fact or law.
- 4.3 <u>Binding Effect of Rulings</u>. Any ruling, regulation, procedure or decision of the Administrator, including any interpretation of the Plan, which is made in good faith shall be conclusive and binding upon all persons affected by it. There shall be no appeal from any ruling by Administrator, except as provided in Section 4.4 below. When making a determination or a calculation, the Administrator shall be entitled to rely on information supplied by investment managers, insurance institutions, accountants and other professionals including legal counsel for the Administrator. Any rule or procedure established by the Administrator may alter any provision of this Plan that is ministerial or procedural in nature without the necessity for a formal amendment of the Plan.

4.4 Claims Procedure.

- (a) Any Participant or Beneficiary, or any other person asserting the right to receive a benefit under this Plan by virtue of his or her relationship to a Participant or Beneficiary (the "Claimant"), who believes that he or she has the right to a benefit that has not been paid, must file a written claim for such benefit in accordance with the procedures established by the Administrator. All such claims shall be filed not more than one year after the Claimant knows, or with the exercise of reasonable diligence would have known, of the basis for such claim. The preceding sentence shall not be construed to require a Participant or Beneficiary to file a formal claim for the payment of undisputed benefits in the normal course, but any claim that relates to the amount of any benefit shall in any event be filed not more than one year after payment of such benefit commences. The Administrator may retain third party administrators and recordkeepers for the purpose of processing routine matters relating to the payment of benefits, but correspondence between a Participant, Beneficiary or other person and such third parties shall not be considered claims for purposes of this Section, and a person shall not be considered a Claimant until he or she has filed a written claim for benefits with the Administrator.
- (b) All claims for benefits shall be processed by the Administrator, and the Administrator shall furnish the Claimant within 90 days after receipt of such claim a written notice that specifies the reason for the denial, refers to the pertinent provisions of the Plan on which the denial is based, describes any additional material or information necessary for properly completing the claim and explains why such material or information is necessary, and explains the claim review procedures of this Section 4.4, and the Claimant's right to bring an action under Section 502 of ERISA, subject to the restrictions of paragraph (e) if the request for review is unsuccessful. The 90 day period may be extended by up to an additional 90 days if the Administrator so notifies the Claimant prior to the end of the initial 90 day period, which notice shall include an explanation of

the reason for the extension and an estimate of when the processing of the claim will be complete. If the Administrator determines that additional information is necessary to process the claim, the Claimant shall be given a period not less than 45 days to furnish the information, and the time for responding to the claim shall be tolled during the period of time beginning on the date on which the Claimant is notified of the need for the additional information and the day on which the information is furnished (or if earlier the end of the period for furnishing the information).

- (c) If the claim is denied in whole or in part, or if the decision on the claim is otherwise adverse, the Claimant may, within 60 days after receipt of such notice, request a review of the decision in writing. If the claimant requests a review, the Employee Benefits Committee (or such other fiduciary as the Administrator may appoint for such purpose) shall review such decision. The Employee Benefits Committee's decision on review shall be in writing and furnished not more than five days after the meeting at which the review is completed, and shall include specific reasons for the decision, written in a manner calculated to be understood by the Claimant, shall include specific references to the pertinent provisions of the Plan on which the decision is based, and shall advise the Claimant of his or her right to bring an action under Section 502 of ERISA, subject to the limitations of paragraph (e).
- (d) The Employee Benefits Committee shall complete its review of the claim not later than its first meeting that is held at least 30 days after the request for review is received. If special circumstances require, the decision may be made by the Employee Benefits Committee not later than its third meeting held after the request for review is received, in which event the Claimant shall be notified of the reason for the delay not later than five days after the meeting at which the review would otherwise have been completed, which notice shall explain the reason for the delay and include an estimate of the time at which the review will be complete. Notwithstanding the foregoing, if at any time the Employee Benefits Committee (or any other fiduciary designated to review appeals) is not scheduled to meet at least quarterly, the decision on review shall be delivered to the Claimant not more than 60 days after the request for review is received, which may be extended to not more than 120 days if special circumstances require and the notice of extension described above is furnished by the end of the initial 60 day period.
- (e) No action at law or in equity shall be brought to recover benefits under this Plan until the claim and appeal rights herein provided have been exercised and the Plan benefits requested in such claim and appeal have been denied in whole or in part. After exhaustion of the Plan's claim procedures, any further legal action taken against the Plan or its fiduciaries by a claimant must be filed in a court of law no later than 120 days after the final adverse benefit determination of the Employee Benefits Committee (or other final appeals fiduciary) is communicated to the claimant or his or her legal representative, notwithstanding any other statute of limitations. In the event a claimant wishes to bring a legal action against the Plan or one of its fiduciaries, such legal action must be filed in the United States District Court for the Northern District of Illinois (Eastern Division) and shall be governed by the procedural and substantive laws of the State of Illinois, to the extent such laws are not preempted by ERISA, notwithstanding any conflict of laws principles.
- (f) The provisions of this Section are intended to comply with ERISA Section 503 and the Department of Labor regulations issued pursuant thereto, and shall be so construed and

applied. Consistent with such regulations, each Claimant shall have the right to have an authorized representative act on his or her behalf, to submit arguments and information in support of his or her claim, and to receive, upon written request and without charge, copies of all documents, records, or other information that either (i) were relied upon in determining his or her benefit under the Plan, (ii) were submitted, considered, or generated in the course of making the benefit determination, even if not relied upon, or (iii) demonstrate compliance with the administrative processes and safeguards of the claim and review procedure.

4.5 <u>Indemnity</u>. To the extent permitted by applicable law and to the extent that they are not indemnified or saved harmless under any liability insurance contracts, any present or former officers, Employees or directors of the Company, and each of them shall be indemnified and saved harmless by the Company from and against any and all liabilities or allegations of liability to which they may be subjected by reason of any act done or omitted to be done in good faith in the administration of the Plan, including all expenses reasonably incurred in their defense in the event that the Company fails to provide such defense after having been requested in writing to do so.

ARTICLE V AMENDMENT AND TERMINATION OF PLAN

- 5.1 <u>Amendment</u>. The Company may amend the Plan at any time by action of the Board, or any person to whom the Board may delegate such authority, except that no amendment shall decrease the vested Account balance of any Participant as of the effective date of the amendment. The Board has delegated the authority to amend the Plan, with certain exceptions, to the Executive Vice President and Chief Human Resources Officer of Continental Casualty Company, and any amendment executed by such officer shall be binding on all parties. In addition, the Administrator is authorized pursuant to Section 4.3 to adopt rules and procedures that have the effect of amendment technical, administrative or ministerial provisions of the Plan. By their execution of this amendment and restatement of the Plan, each Employer ratifies and accepts all prior amendments to the Plan, and agrees that in the future the Plan may be amended by action of the Company without consent of the other Employers.
- 5.2 <u>Termination</u>. The Company may at any time terminate the Plan by action of the Board. Upon termination, no further allocations shall be made to Accounts, but Accounts shall continue to be credited with earnings and shall be paid in accordance with the provisions of the Plan; provided, however, that upon termination, the Company may, but shall not be obligated to, amend the Plan to provide that the Accounts of some or all Participants shall be fully vested and paid to such Participants in a lump sum, which shall fully discharge all obligations owed to such Participants under the Plan; provided that such amendment shall apply to the Post-2004 Accounts only if all such Accounts are fully vested and distributed and the amendment otherwise complies with the requirements of §409A of the Code. Any Employer may at any time withdraw from the Plan by written notice to the Administrator, in which event the Plan shall be considered terminated with respect to the Participants employed by such Employer (or who were so employed at the time of their termination of employment), and the provisions of this Section 5.2 shall apply to such Participants only.

ARTICLE VI MISCELLANEOUS

- 6.1 <u>Status of Plan</u>. This Plan is intended to be an unfunded plan maintained primarily to provide retirement benefits for a select group of management Employees or highly compensated Employees within the meaning of Section 201(1), Section 301(a)(3), and §401(a)(1) of ERISA and Department of Labor Regulations 29 C.F.R. Section 2520.104-23, and shall be so construed.
- 6.2 <u>Nonassignability</u>. Neither a Participant nor any other person shall have any right to commute, sell, assign, transfer, pledge, anticipate, mortgage or otherwise encumber, transfer, hypothecate or convey in advance of actual receipt the amounts, if any, payable hereunder, or any part thereof, which are, and all rights to which are, expressly declared to be nonassignable and nontransferable. No part of the amounts payable shall, prior to actual payment, be subject to garnishment, seizure or sequestration for the payment of any debts owed by a Participant or any other person, nor be transferable by operation of law in the event of a Participant's or any other person's bankruptcy or insolvency. Nothing contained herein shall be construed as a waiver of the Company's or any Employer's right of setoff.
- 6.3 <u>No Contract of Employment</u>. The terms and conditions of this Plan shall not be deemed to constitute a contract of employment between the Company or any Employer and the Participant, and neither the Participant nor the Participant's Beneficiary shall have any rights against the Company or any Employer except as may otherwise be specifically provided herein. Moreover, nothing in this Plan shall be deemed to give a Participant the right to be retained in the service of the Company or any Employer or to interfere with the right of the Company and each Employer to discipline or discharge him at any time.
- 6.4 Participant Litigation. In any action or proceeding regarding the Plan, Participants, Employees or former Employees of the Company or an Employer, their Beneficiaries or any other persons having or claiming to have an interest in this Plan shall not be necessary parties and shall not be entitled to any notice or process. Any final judgment which is not appealed or appealable and may be entered in any such action or proceeding shall be binding and conclusive on the parties hereto and all persons having or claiming to have any interest in this Plan. To the extent permitted by law, if a legal action is begun against the Company, an Employer, the Administrator, the trustee of any trust established hereunder, or any person acting on the behalf or under the direction of any of the foregoing persons, by or on behalf of any person and such action results adversely to such person or if a legal action arises because of conflicting claims to a Participant's or other person's benefits, the costs to any such person of defending the action will be charged to the amounts, if any, which were involved in the action or were payable to the Participant or other person concerned. To the extent permitted by applicable law, acceptance of participation in this Plan shall constitute a release of the Company, each Employer, the Administrator and such trustee and their respective agents from any and all liability and obligation not involving willful misconduct or gross neglect.

- 6.5 <u>Participant and Beneficiary Duties</u>. Persons entitled to benefits under the Plan shall file with the Administrator from time to time such person's post office address and each change of post office address. Each such person entitled to benefits under the Plan also shall furnish the Administrator with all appropriate documents, evidence, data or information which the committee considers necessary or desirable in administering the Plan.
- 6.6 Governing Law. The provisions of this Plan shall be construed and interpreted according to the laws of the State of Illinois to the extent not pre-empted by the laws of the United States. The Plan is intended to comply with all requirements of §409A, and to the maximum extent permitted by law shall be construed in a manner that is consistent with such intent, provided that in no event shall the Company, any Employer, the Administrator, or any of their respective employees, agents or representatives have any liability to any person for any additional tax imposed upon such person pursuant to §409A of the Code or any comparable federal, state or local income tax law.
- 6.7 <u>Validity</u>. In case any provision of this Plan shall be held illegal or invalid for any reason, such illegality or invalidity shall not affect the remaining parts hereof, but this Plan shall be construed and enforced as if such illegal and invalid provision had never been inserted herein.
- 6.8 <u>Notices</u>. Any notice or filing required or permitted to be given to the Administrator or the Company under the Plan shall be sufficient if in writing and hand delivered, or sent by registered or certified mail to the Company at its principal executive offices, or to Company's statutory agent. Notices shall be deemed given as of the date of delivery or, if delivery is made by mail, as of the date shown on the postmark on the receipt for registration or certification. Any notice required or permitted to be given to a Participant shall be sufficient if in writing and hand delivered or sent by first class mail to the Participant at the last address listed on the records of the Company or such Participant's Employer.
- 6.9 <u>Successors</u>. The provisions of this Plan shall bind and inure to the benefit of each Employer and its respective successors and assigns. The term successors as used herein shall include any corporate or other business entity which shall, whether by merger, consolidation, purchase or otherwise acquire all or substantially all of the business and assets of an Employer, and successors of any such corporation or other business entity.

IN WITNESS WHEREOF, the Company ha	s caused this amendment and restatement of	of the Plan to be executed on	March 29,	2018
	CNA FINANCIAL CORPORATION			

By: /s/ Elizabeth Aguinaga

Elizabeth Aguinaga Executive Vice President & Chief Human Resources Officer, Continental Casualty Company

APPENDIX A

FULL VESTING OF PARTICIPANTS AFFECTED BY CERTAIN EVENTS

A.1 Sales of Business Units

In accordance with Section 2.5, Participants whose employment is terminated in connection with the following sales or other dispositions of business units shall be fully vested in their Account balance regardless of their years of service. Except as otherwise provided below, the Participants who qualify for full vesting with respect to any transaction shall be those, and only those, who qualify as an "Affected Member" with respect to such transaction in accordance with Appendix F of the 401(K) Plus Plan.

Transaction	Closing Date	Exceptions/Special Rules
Sale of Life Reinsurance Business Unit to MARC	12/31/00	None
Sale of CNA Credit Collection Agency, Inc., to Coface	12/31/02	None
Sale of the unbundled risk management business of RSKCo Services, Inc to Cunningham Lindsey US	6/2/03	None
Sale of Smith System to McFadden Brothers	4/29/03	None
Sale of CNA Group Operations to Hartford Financial Services Group	12/31/03	None
Sale of individual life insurance business to Swiss Re Life & Health America	App. 3/31/04	None

APPENDIX B EXCESS BENEFIT PLAN

- B1. Purpose and Interpretation. The purpose of this Appendix B is to establish a separable portion of the Plan, referred to herein as the "Excess Benefit Plan", solely for the purpose of providing benefits for Employees whose benefits under the 401(K) Plus Plan are restricted by Code §415. Except as otherwise specifically provided in this Appendix B, all provisions of the Plan shall apply to Participants in the Excess Benefit Plan to the same extent and in the same manner such provisions apply to Participants in the Plan. All capitalized terms used in this Appendix B shall have the same meaning as such terms as defined in the Plan. It is the Company's intent that the benefits accrued for Participants solely under this Appendix B shall constitute a separable part of the Plan that constitutes an "excess benefit plan" as defined in §3(36) of ERISA.
- B2. <u>Eligibility</u>. The Employees who are eligible to participate in the Excess Benefit Plan shall be those Employees whose benefits under the 401(K) Plus Plan are restricted solely by Code §415, and who are not, and never have been, eligible to participate in the Plan pursuant to Section 2.1 thereof. An Employee shall become a Participant in the Excess Benefit Plan on the first payroll date on which an amount that would otherwise have been credited to his account in the 401(K) Plus Plan cannot be credited solely by reason of Code §415, and shall remain a Participant in the Excess Benefit Plan until either his Account is fully distributed, or he transfers to the Plan pursuant to Section B4 below.
- B3. Benefits Credited to Account. The amount credited to the Account of a Participant in the Excess Benefit Plan in each Plan Year shall be equal to the amount that would have been credited to his Account in the Plan pursuant to Section 2.3 (other than the first sentence of Section 2.3(a)) for such Plan Year if he were eligible to participate in the Plan for such Plan Year, but taking into account only the Tax Limit imposed by Code §415. A Participant's Account in the Excess Benefit Plan shall be treated as an Account in the Plan for all purposes of the Plan.
- B4. <u>Payment Election</u>. When an Employee first becomes eligible to participate in the Excess Benefit Plan, but not more than 30 days after such date, the Administrator may permit the Employee to make an election as to the form of payment of his benefit in accordance with Section 2.6(b) (provided that the Employee is not described in the provision to Section 2.6(b)(ii)), which election shall thereafter be treated as an election made under Section 2.6(b).
- B5. <u>Transfer to Plan</u>. If an Employee who has been a Participant in the Excess Benefit Plan in any Plan Year subsequently becomes eligible to participate in the Plan pursuant to Section 2.1, his Account under the Excess Benefit Plan shall be transferred to the Plan, and he shall thereafter be considered a Participant in the Plan and shall be ineligible to again participate in the Excess Benefit Plan for so long as he remains an Employee, even if he subsequently becomes ineligible to participate in the Plan.

SARBANES-OXLEY ACT SECTION 302

CERTIFICATION OF CHIEF EXECUTIVE OFFICER

I, Dino E. Robusto, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of CNA Financial Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth quarter in the case of an Annual Report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officers and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent function):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Dated:	April 30, 2018	Ву	/s/ Dino E. Robusto	
			Dino E. Robusto	
		Chief Executive Officer		

SARBANES-OXLEY ACT SECTION 302

CERTIFICATION OF CHIEF FINANCIAL OFFICER

I, D. Craig Mense, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of CNA Financial Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth quarter in the case of an Annual Report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officers and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent function):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Dated:	April 30, 2018	Ву	/s/ D. Craig Mense
			D. Craig Mense
			Chief Financial Officer

Written Statement of the Chief Executive Officer

of CNA Financial Corporation

Pursuant to 18 U.S.C. § 1350

(As adopted by Section 906 of the Sarbanes-Oxley Act of 2002)

The undersigned, the Chief Executive Officer of CNA Financial Corporation (the Company), hereby certifies that, to his knowledge:

- the Company's quarterly report on Form 10-Q for the quarter ended March 31, 2018 filed on the date hereof with the Securities and Exchange Commission (the Report) fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated:	April 30, 2018	Ву	/s/ Dino E. Robusto
		_	Dino E. Robusto
			Chief Executive Officer

The foregoing certification is being furnished solely pursuant to 18 U.S.C. § 1350 and is not being filed as part of the Report or as a separate disclosure document.

Written Statement of the Chief Financial Officer

of CNA Financial Corporation

Pursuant to 18 U.S.C. § 1350

(As adopted by Section 906 of the Sarbanes-Oxley Act of 2002)

The undersigned, the Chief Financial Officer of CNA Financial Corporation (the Company), hereby certifies that, to his knowledge:

- the Company's quarterly report on Form 10-Q for the quarter ended March 31, 2018 filed on the date hereof with the Securities and Exchange Commission (the Report) fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated:	April 30, 2018	By	/s/ D. Craig Mense
		_	D. Craig Mense
			Chief Financial Officer

The foregoing certification is being furnished solely pursuant to 18 U.S.C. § 1350 and is not being filed as part of the Report or as a separate disclosure document.